This tariff, Idaho Price List No. 2 submitted on behalf of XO Communications Services, Inc., cancels and replaces Idaho Intrastate Access Services Price List, submitted on behalf of XO Idaho, Inc.

XO Communications Services, Inc.

Regulations, Descriptions, and Rates

Applicable to Furnishing Intrastate Access Services

Within the State of Idaho

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005

CHECK SHEET

Pages of this tariff, as indicated below, are effective as of the date shown at the bottom of the respective pages. Original and revised pages, as named below, comprise all changes from the original tariff and are currently in effect as of the date on the bottom of this page.

PAGE	REVISION		PAGE	REVISION	PAGE	REVISION	
Title	Original		26	Original	51	Original	
1	22 nd Rev.	*	27	1st Rev.	52	1st Rev.	
2	1st Rev.		28	1st Rev.	53	Original	
3	1st Rev.		29	Original	54	Original	
4	5th Rev.		30	Original	54.1	Original	
5	Original		31	Original	54.2	Original	
6	Original		32	Original	54.3	Original	
7	3 rd Rev.		33	Original	54.4	Original	
8	4th Rev.		34	Original	54.5	Original	
9	Original		35	Original	55	Original	
10	1st Rev.		36	1st Rev.	56	Original	
11	Original		37	Original	57	Original	
12	Original		38	Original	58	Original	
13	Original		39	Original	59	Original	
14	Original		40	Original	60	Original	
15	Original		41	Original	61	Original	
16	Original		42	Original	62	Original	
17	Original		43	Original	63	Original	
18	Original		44	Original	64	2 nd Rev.	
19	Original		45	2nd Rev.	65	1st Rev.	
20	1st Rev.		46	1st Rev.	66	6th Rev.	*
21	Original		47	2 nd Rev.	67	3 rd Rev.	*
22	1st Rev.		47.1	Original	68	1st Rev.	
22.1	4th Rev.		48	1st Rev.	69	Original	
22.2	3 rd Rev.		49	Original	70	2nd Rev.	
22.3	1st Rev.		50	Original	70.1	5th Rev.	*
22.4	1st Rev.				70.2	Original	
22.5	2 nd Rev.				71	1st Rev.	
22.6	Original						
23	Original						
24	Original						
25	Original						

Issued: June 15, 2022 Effective: July 1, 2022

Karl Tucker - Director 22001 Loudoun County Pkwy (T) Ashburn, VA 20147 (T)

TABLE OF CONTENTS

	K SHEET		2	
TABLE OF CONTENTS			3	
EXPLANATION OF SYMBOLS			5	
APPLI	CATION	OF PRICE LIST	6	
1.	DEFI	NITIONS	7	
2.	REGU	ILATIONS	9	
2.1	Under	taking of the Company	9	
<u>د. ۱</u>	2.1.1	Shortage of Equipment or Facilities	9	
	2.1.1	Terms and Conditions	9	
	2.1.2	Liability and the Company	ĺI	
	2.1.4	Notification of Service-Affecting Activities	14	
	2.1.5	Provisions of Equipment and Facilities	15	
	2.1.6	Non-Routine Installation	16	
	2.1.7	Special Construction	16	
	2.1.8	Ownership of Facilities	17	
2.2		ited Uses	17	
2.3		ation of the Customer	17	
2.4		ner Equipment and Channels	22	
	2.4.1	In General	22	
	2.4.2	Station Equipment22		
	2.4.3	Interconnection of Facilities	22	
	2.4.4	Inspections and Testing	23	
2.5		ent Arrangements	23	
	2.5.1	Payment for Service	23	
	2.5.2	Billing and Collection of Charges	24	
	2.5.3	Billing or Payment Disputes	27	
	2.5.4	Advance Payments	29	
	2.5.5	Deposits	29	
	2.5.6	Discontinuance of Service	30	
	2.5.7	Cancellation of Application for Service	33	
	2.5.8	Changes in Service Requested	33	
2.6	Allow	ance for Interruptions in Service	34	
2.6.1	2.6.1	Credit for Interruptions	34	
	2.6.2	Limitations on Allowances	35	(72)
				(D)
2.7		llation of Service	36	
2.8		ers and Assignments	36	
2.9	Notice	s and Communications	37	

Issued: June 7, 2013

Kelly Faul – Regulatory Affairs Director 13865 Sunrise Valley Dr. Herndon, VA 20171 Effective: June 30, 2013
Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

TABLE OF CONTENTS, (CONT'D.)

3.		ING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE	37	
3.1	General		37	
	3.1.1	Ordering Conditions	37	
2.2	3.1.2	Provision of Other Services	38	
3.2	Access C		38	
	3.2.I	Access Service Date Intervals	39	
	3.2.2	Access Service Request Modifications	40	
	3.2.3	Cancellation of an Access Service Request	42	
	3.2.4	Minimum Period of Service	43	
4.	SPECIA	L ACCESS	44	
	4.1	General	44	
	4.2	Transmission Service	44	
5.	SWITCH	HED ACCESS SERVICE	45	
5. 5.1	General	IED ACCESS SERVICE	45	
5.1		n and Description of Switched Access Service Agreements	45	
3.2	5.2.1	Feature Group D (FG D) Access	45 45	
	5.2.1	Manner of Provision		
			46	
	5.23.	Rate Categories	46	
	5.2.4	Billing Validation	49	
	5.25.	Design Layout Report	49	
	5.26.	Acceptance Testing	50	
	5.2.7	Ordering Options and Conditions	50	
	5.2.8	Competitive Pricing Arrangement	50	
5.3		ons of the Company	50	
	5.3.1	Network Management	50	
5.4		ons of the Customer	51	
	5.4.1	Report Requirements	51	
	5.4.2	Supervising Signaling	51	
	5.4.3	Design of Switched Access Services	51	
5.5	Switched	Access Optional Features	52	
	5.5.1	Nonchargeable Optional Features	52	
	5.5.2	Chargeable Optional Features		
	5.5.3	Feature Group D Optional Features	52	
5.6	Switched	Access IP Direct Connect	54.1	(N)
	5.6.1	Provision of Switched Access IP Direct Connect	54.1	
	5.6.2	Eligibility	54.2	
	5.6.3	Expiration Date	54.2	
	5.6.4	Billing of SWA IP Direct Connect Service Minutes	54.2	
	5.6.5	General Regulations	54.3	
	5.6.6	Obligations of Company	54.4	
	5.6.7	Obligations of the Customer	54.5	
			2 7,0	(N

Issued: October 5, 2012

Kelly Faul – Regulatory Affairs Director 13865 Sunrise Valley Dr. Herndon, VA 20171 Effective: October 15, 2012
Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

OCT 1 5 2012

TABLE OF CONTENTS, (CONT'D.)

6.	RATES AND CHARGES	55
6.1	Special Construction	55
	6.1.1 Basis for Rates and Charges	55
	6.1.2 Basis for Cost Computation	55
	6.1.3 Termination Liability	56
6.2	Special Access Service	57
	6.2.1 General	57
	6.2.2 Voice Grade Service	57
	6.2.3 Digital Data Services	59
	6.2.4 DS -1 Service	61
	6.2.5 DS - 3 Service	62
	6.26. Service Calls	63
	6.2.7 Individual Case Basis Arrangement	63
6.3	Switched Access Service	64
	6.3.1 Application of Rates	64
	6.3.2 Billing of Access Minutes	4
	6.3.3 Rates and Charges	6
	6.3.4 Special Construction	67
	6.3.5 Billing Name and Address Service	68
6.4	Access Service Order Charge	70
6.5	Historic Invoices	70
6.6	Service Order Changes	70
6.7	Intra-Building Moves	70.1
6.8	Switched Access IP Direct Connect	70.1
6.9	Service Extension Charge	70.2 (N)
7.	INTERIM INTERCONNECTION	71
8.	RESERVED FOR FUTURE USE	71

Issued: October 10, 2013

Kelly Faul – Regulatory Affairs Director 13865 Sunrise Valley Dr. Herndon, VA 20171 Effective: October 30, 2013

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

EXPLANATION OF SYMBOLS, REFERENCE MARKS, AND ABBREVIATIONS OF TECHNICAL TERMS USED IN THIS PRICE LIST.

The following symbols shall be used in this price list for the purpose indicated below:

- (C) To signify changed conditions or regulation.
- (D) To signify discontinued rate, regulation or condition.
- (I) To signify increase.
- (K) To signify that material has been transferred to another sheet or place in the price list.
- (M) To signify that material has been transferred from another sheet or place in the price list.
- (N) To signify a new rate, regulation, condition or sheet.
- (O) To signify no change.*
- (R) To signify reduction.
- (T) To signify a change in text for clarification.
- * The use of the symbol "O" shall be discretionary unless its use in the interest of clarity is evident or specifically requested by the Commission.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005

APPLICATION OF PRICE LIST

This Price List sets forth the service offerings, rates, terms and conditions applicable to the furnishing of intrastate switched and special access within the State of Idaho by XO Communications Services, Inc. hereinafter referred to as the Company. Nothing contained herein shall, however, prohibit or restrict the right of XO to enter into contracts with customers that may contain rates, terms, and conditions different from those contained herein.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005 Effective: March 13, 2005

1. DEFINITIONS

8YY Originating – Feature Group D originating usage dialed with the following NPAs: 800, 888, 877, 866, 855, 844, 833, 822, or other NPAs as they become available to the industry for the provision of Toll Free services.

(N) (N)

(T)

(T)

(N)

(N)

Access: A connection between a Customer Premises and a Point of Presence of an Interchange Company for the transmission of voice, data or video/image information.

Advance Payment: Part of all of a payment required before the start of service.

<u>Automatic Number Identification (ANI)</u> – The Multi-Frequency (MF) signaling parameter that identifies the billing number of the calling party.

Bit: The smallest unit of a payment required before the start of service.

<u>Calling Party Number (CPN)</u> – The SS7 signaling parameter that identifies the subscriber line number or directory number of the calling party.

Charge Number (CN) - The SS7 signaling parameter that identifies the billing telephone number of the calling party.

Commission: The Idaho Public Service Commission

<u>Company:</u> XO Communications Services, LLC, the issuer of this Price List, which is a Delaware limited liability company.

<u>Customer:</u> The person, firm, corporation, government entity, or other entity which subscribes to or uses, and is therefore responsible for the payment of charges and compliance with the Company's regulations, the services offered under this tariff, including both Interexchange Carriers (IXCs) and End Users and other carriers or providers that originate or terminate Toll VoIP-PTSN Traffic, or otherwise exchange Toll VoIP-PSTN Traffic with the Company.

<u>Dedicated:</u> A facility or equipment system or subsystem set aside for the sole use of a specific Customer.

<u>Duplex Service</u>: Service which provides for simultaneous transmission in both directions.

<u>End User</u>: Any customer of an intrastate communications service that is not a common carrier; provided that a common carrier other than a telephone company shall be deemed to be an "end user" when such common carrier uses a communications service for administrative purposes, and a person or entity that offers communications service exclusively as a reseller shall be deemed to be an "end user" if all resale transmissions offered by such reseller originate on the premises of such reseller.

Entrance Facility - An Entrance Facility provides the communication path between a Customer's premises and the Company for the sole use of the Customer.

<u>Fiber Optic Cable:</u> A thin filament of glass with a protective outer coating through which a light beam carrying communications signals may be transmitted by means of multiple internal reflections to a receiver, which translates the message.

<u>Individual Case Basic (ICB):</u> Denotes service provided and performed by the Company involving special engineering, design, programming, development, or production activities to provide services not otherwise proved under this Price List. Rates and charges are developed based on the specific circumstances of the case.

<u>InterLATA Service:</u> Service which originates within one Local Access Area Transport Area (LATA) and terminates in a different LATA.

Internet Protocol (IP) Signaling - A packet data-oriented protocol used for communications call signaling information.

IntraLATA Service: Service which originates and terminates within the same Local Access Transport Area (LATA)

Local Access: Local Access means the connection between a Customer premises and Company Point of Presence.

Mbps: Megabits per second.

(Some material previously located on this page is now on page 8.)

Issued: June 17, 2021 Effective: July 1, 2021

Kelly Faul - Senior Manager (T) 22000 AOL Way (T)

Dulles, VA 20166 (T)

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING
July 1, 2021
Boise, Idaho

1. <u>DEFINITIONS</u> (Cont'd.)

<u>Multi-Frequency (MF) Signaling</u> - An in-band signaling method in the which call signaling information is transmitted between network switches using the same voice band channel used for voice.

Network: The Company's digital fiber optics-based network.

Network Services: The Company's telecommunications access services offered on the Company's Network.

<u>Node:</u> The Company office where all Customer facilities are terminated for purposes of interconnection to trunks and/or cross-connection to distant ends.

Non-8YY Originating – Feature Group B or D originating usage that is not dialed with the following NPAs: 800, 888, 877, 866, 855, 844, 833, 822, or other NPAs as they become available to the industry for the provision of Toll Free services.

Non-Recurring Charges: The one-time initial charges for services or facilities, including but not limited to charges for construction, installation, or special fees, for which the Customer becomes liable at the time of Service Order is executed.

PIU: Percent Interstate Usage

<u>Point to Point Service</u>: Point to Point Service is an unswitched full time transmission service utilizing the Company's facilities to connect two or more Customer designated locations.

Premises: Denotes a building or buildings on contiguous property (except railroad right-of-way, etc.) not separated by a public highway.

Rate Zone - A Rate Zone is a geographic area that comprises a specific Incumbent Local Exchange Carrier's (ILEC's) operating territory within the state. All usage rates and charges will be assessed based on the operating territory of the Incumbent Local Exchange Carrier ("ILEC") from which a call originates or to which a call terminates. Zone 1 is the operating territory of CenturyLink Operating Companies (Tariff FCC No. 11). Zone 2 is the operating territory of Ziply Fiber Telephone Companies (Tariff FCC No. 2). For those calls that originate or terminate in any area of the state not within an identified zone, the Zone 1 rates will apply.

<u>Recurring Charges:</u> The monthly charges to the Customer for services, facilities and equipment, which continue for the agreed upon duration of the service.

Service Commencement Date: The first day following the date on which the Company notifies the Customer that the requested service or facility is available for use, unless extended by the Customer's refusal to accept service which does not conform to standards set forth in the Service Order or this Price List, in which case the Service Commencement Date is the date of the Customer's acceptance. The parties may

mutually agree on a substitute Service Commencement Date. In any case, the Service Commencement Date is no later than the date that the Customer uses the Company's Service.

<u>Service Order:</u> The written request for Network Services executed by the Customer and The Company in the format devised by the Company or the use of Company Services by the Customer. The signing of a Service Order by the Customer and acceptance by the company or the use of Company Services by the Customer initiates the respective obligations of the parties as set forth therein and pursuant to this Price List, but the duration of the service calculated from the Service Commencement Date.

<u>Service</u>: The Company's telecommunications access service offered on the Company's network and provided to the Customer by the Company either pursuant to a written request for Network Services or through the Customer's use of the Company's services.

Shared: A facility or equipment system or sub system which can be used simultaneously by several Customers.

Toll Free - Toll Free is a service for which the toll charges for completed calls are paid by the toll free subscriber, rather than the calling party.

Toll VoIP-PSTN Traffic —Interexchange voice traffic that (1) is exchanged with the Company in Time Division Multiplexing ("TDM") format over Public Switched Telephone Network ("PSTN") facilities and (2) originates and/or terminates in Internet Protocol ("IP") format. Toll VoIP-PSTN Traffic originates and/or terminates in IP format when it originates from and/or terminates to an end user of a service that requires IP-compatible customer premises equipment.

Issued: June 17, 2021 Effective: July 1, 2021

Kelly Faul - Senior Manager (T) 22000 AOL Way (T) Dulles, VA 20166 (T)

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING
July 1, 2021
Boise, Idaho

(N)

(N)

(N)

(N)

(N) (N)

2. REGULATIONS

2.1 <u>Undertaking of the Company</u>

Access Services consist of furnishing communications service in connection with one way and/or two-way information transmission between points within Idaho, under the terms of this Price List.

2.1.1 Shortage of Equipment or Facilities

- 2.1.1.1 The Company reserves the right to limit or to allocate the use of existing facilities, or of additional facilities offered by the Company, when necessary because of lack of facilities, or due to some other cause beyond the Company's control.
- 2.1.1.2 The furnishing of service under this Price List is subject to the availability on a continuing basis of all the necessary facilities and is limited to the capacity of the Company's fiber optic cable facilities as well as facilities the Company may obtain from other carries to furnish service from time to time as required at the sole discretion of the Company.
- 2.1.1.3 The provisioning and restoration of services in emergencies shall be in accordance with Part 64, Support D, Appendix A, of the Federal Communications Commission's Rules and Regulations, which specifies the priority system for such activities.

2.1.2 <u>Terms and Conditions</u>

- 2.1.2.1 Except as otherwise provided herein, service is provided on the basis of a minimum period of at least one month and shall continue to be provided until canceled by the Customer, in writing, on not less than 30 days notice. For the purpose of computing charges in this Price List, a month is considered to have 30 days.
- 2.1.2.2 Customers may be required to enter into written Service Orders which shall contain or reference the name of the Customer, a specific description of the service ordered, the rates to be charged, the duration of the services, and the terms and conditions in this Price List. Customer will also be required to execute any other documents as may

be reasonably requested by the Company. If the Customer uses the Services, but does not execute a written Service Order, the Customer is deemed to have ordered the Services and is obligated to pay for the Services.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Effective: March 13, 2005

Issued: March 3, 2005

2. <u>REGULATIONS</u> (Cont'd)

- 2.1 <u>Undertaking of the Company</u> (Cont'd)
 - 2.1.2 <u>Terms and Conditions</u> (Cont'd)
 - 2.1.2.3 Service may be terminated at the end of the term by either party providing the other with written notice of termination at least thirty (30) days prior to the termination date. If the customer does not cancel Service before the end of the Initial Term, XO will automatically renew Service for a similar term at rates set forth in this tariff. Any termination shall not relieve the Customer of its obligation to pay any charges incurred for Services used or ordered under this tariff. The rights and obligations which by their nature extend beyond the termination of the term of the Service order shall survive such termination.
 - 2.1.2.4 In the event the Company incurs fees or expenses, including without limitation attorneys' fees, in collecting, or attempting to collect, any charges owed the Company, the Customer will be liable to the Company for the payment of all such fees and expenses reasonably incurred.
 - 2.1.2.5 Service may be terminated upon written notice to the customer if:
 - a. the Customer is using the service in violation of this Price List, or
 - b. the Customer is using the service in violation of the law.
 - 2.1.2.6 This Price List shall be interpreted and governed by the laws of the State of Idaho without regard for its choice of laws provision.
 - 2.1.2.7 The Customer agrees to operate Company-provided equipment in accordance with instructions of the Company or its agent. Failure to do so will void company liability for interruption of service and may make the customer responsible for damage to equipment pursuant to Section 2.1.2.8 below.
 - 2.1.2.8 The Customer agrees to return to the company all company-provided equipment delivered to the Customer within five (5) days of termination of the service in connection with which the equipment was used. Said equipment shall be in the same condition as when delivered to the customer, except for normal wear and tear.

Customer agrees to reimburse the Company, upon demand, for any reasonable costs incurred by the Company due to the Customer's failure to comply with this provision.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

NOV 5 - 2010

Boise, Idaho

Effective: November 5, 2010

Issued: October 26, 2010

Kelly Faul – Regulatory Affairs Director

13865 Sunrise Valley Dr. Herndon, VA 20171

2. REGULATIONS (Cont'd)

- 2.1 Undertaking of the Company (Cont'd)
 - Liability of the Company (Cont'd) 2.1.3
 - The liability of the Company for damages arising out of the furnishing of its Services, including but not limited to mistakes, omissions, interruptions, delays, or errors, or other defects, representations, or use of the service or arising out of the failure to furnish the service, whether caused by acts or omission, shall be limited to the extension of allowance for interruption as set forth in 2.6. The extension of such allowances for interruption shall be the sole remedy of the Customer and the sole liability of the Company. The Company will not be liable for any direct, indirect, incidental, special, consequential, exemplary or punitive damages to Customer as a result of any Company service, equipment or facilities, or the acts or omissions or negligence of the Company's employees or agents.
 - The Company shall not be liable for any delay or failure of performance or 2.1.3.2 equipment due to causes beyond its control, including but not limited to: acts of God, fire, flood, explosion or other catastrophes; any law, order, regulation, direction, action, or request of the United State Government, or of any other local governments having or claiming.

Jurisdiction over the Company, or of any department, agency, commission, bureau, corporation, or other instrumentality of any one or more of these federal, state or local governments, or of any civil or military authority; national emergencies; insurrections; riots; wars; unavailability of rights-of-way or materials; or strikes, lockouts, work stoppages, or other labor difficulties.

- The Company shall not be liable for (a) any act or Omission of any entity 2.1.3.3 furnishing to the Company or to the Company's Customers facilities or equipment used for interconnection with Network Services; or (b) for the acts or omissions of common carriers or warehousemen.
- The Company shall not be liable for any damages or losses due to the fault or 2.1.3.4 negligence of the Customer or due to the failure or malfunction of Customer provided equipment or facilities.
- The Customer indemnifies and holds the Company harmless from any and all 2.1.3.5 loss, claims, demands, suits, or other reaction, or any liability whatsoever, whether suffered, made, instituted, or asserted by any other party or person(s), and for any loss, damage, or destruction of any property, whether owned by the Customer or others, caused or claimed to have been caused directly or indirectly by the installation, operation, failure to operate, maintenance, removal presence, condition, location, or use of any installation so provided. The Company reserves the right to require each Customer to sign an agreement acknowledging acceptance of the Provisions of this section as a condition precedent to such installations.

Issued: March 3, 2005

Effective: March 13, 2005

Idaho Public Utilities Commission

Alaine Miller, VP - Regulatory & External Affairs ACCEPTED FOR FILING 1633 Westlake Avenue, No., Suite 200 Seattle, WA 98109

IDa0502

2. <u>REGULATIONS</u> (Cont'd)

- 2.1 <u>Undertaking of the Company</u> (Cont'd)
 - 2.1.3 Liability of the Company (Cont'd)
 - 2.1.3.6 The Company is not liable for any defacement of or damage to Customer premises resulting from the furnishing of services or equipment on such Premises or the installation or removal thereof, unless such defacement or damage is caused by negligence or willful misconduct of the Company's agents or employees.
 - 2.1.3.7 Notwithstanding the Customer's obligations as set forth in Section 2.3.2 below, the Company shall be indemnified, defended and held harmless by the Customer, or by others authorized by it to use the service, against any claim, loss or damage arising from Customer's use of services furnished under this Price List, including: claims for libel, slander, invasion of privacy or infringement of copyright arising from the material, data, information, or other content transmitted via the Company's service; and patent infringement claims arising from combining or connecting the service offered by the Company with apparatus and systems of the Customer or others; all other claims arising out of any act or omission of the Customer or others, in connection with any service provided by the Company pursuant to this Price List.
 - 2.1.3.8 The entire liability for any claim, loss, damage or expense from any cause whatsoever shall in no event exceed sums actually paid Company by Customer for the specific services giving rise to the claim. No action or proceeding against the Company shall be commenced more that one year after the service is rendered.
 - 2.1.3.9 THE COMPANY MAKES NO WARRANTIES OR REPRESENTATIONS, EXPRESS OR IMPLIED EITHER IN FACT OR BY OPERATION OF LAW, STATUTORY OR OTHERWISE, INCLUDING WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR USE, EXCEPT THOSE EXPRESSLY SET FORTH HEREIN.
 - 2.1.3.10 The Company's liability for willful misconduct, if established as a result of judicial or administrative proceedings, is not limited by this Price List. The Company's liability, if any, with regard to delayed installation of Company facilities or commencement of service, shall not exceed \$1,000. With respect to any other claim or suit, by a Customer or by any others, for damages associated with the ordering (including the reservation of any specific number for use with a service), installation (including delays thereof), provision, termination, maintenance, repair interruption or restoration of any service or facilities offered under this Price List, and subject to the provisions of Section 2.6, the Company's liability, if any, shall be limited as provided herein.

Issued: March 3, 2005

Idaho Public Utilities Commission
Office of the Secretary
Alaine Miller, VP - Regulatory & External Affairs

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

1633 Westlake Avenue, No., Suite 200 Seattle, WA 98109

MAR 1 3 2005

IDa0502

- 2. REGULATIONS (Cont'd)
 - 2.1 Undertaking of the Company (Cont'd)
 - 2.1.3 Liability of the Company (Cont'd)
 - 2.1.3.11 The Company shall be indemnified and held harmless by the Customer against any claim, loss or damage arising from the Customer's use of services offered under this Price List including: claims for libel, slander, invasion of privacy or infringement of copyright arising from the Customer's or the Customer's End Users own communications; patent infringement claims arising from the Customer's or the Customer's End Users combining or connecting the service offered by the Company with facilities or equipment furnished by the End User of another Interexchange Carrier; or all other claims arising out of any act or omission of the Customer or the Customer's End User in connection with any service provided pursuant to this Price List.
 - 2.1.3.12 The Company does not guarantee nor make any warranty with respect to service installations at locations at which there is present an atmosphere that is explosive, prone to fire, dangerous or otherwise unsuitable for such installations. The Customer and End User shall indemnify and hold the all loss, claims, demands, suits or other Company harmless from any and actions, or any liability whatsoever, whether suffered, made, instituted or asserted by the Customer or by any other party, for any personal injury to, or death of, any person or persons, or for any loss, damage or destruction of any property, whether owned by the Customer or others, caused or claimed to have been caused directly or indirectly, by the installation, operation, failure to operate, maintenance, removal, presence, condition, location or use of service furnished by the Company at such locations.
 - 2.1.3.13 The Company shall not be liable for the Customer's failure to fulfill its obligations to take all necessary steps including, without limitation, obtaining, installing and maintaining all necessary equipment, materials and supplies, for interconnecting the terminal equipment or communications system of the Customer, or any third party acting as its agent, to the Company's Network. The Customer shall secure all licenses, permits, rights-of-way, and other arrangements necessary for such interconnection. In addition, the Customer shall ensure that its equipment and/or system or that of its agent is properly interfaced with the Company's service, that the signals emitted into the Company's Network are of the proper mode, band-width, power, data speed, and signal level for the intended use of the Customer and in compliance with the criteria set forth in Section 2.1.6 following, and that the signals do not damage Company equipment, injure its personnel or degrade service to other Customers. If the Customer or its agent fails to maintain and operate its equipment and/or system or that of its agent properly, with resulting imminent harm to Company equipment, personnel, or the quality of service to other Customers, the Company, may, upon written notice, require the use of protective equipment at the Customer's expense.

Issued: March 3, 2005

Idaho Public Utilities Commission Live: March 13, 2005 Office of the Secretary

Alaine Miller, VP - Regulatory & External AGGEPTED FOR FILING 1633 Westlake Avenue, No., Suite 200 Seattle, WA 98109

MAR 1 3 2005

IDa0502

2. <u>REGULATIONS</u> (Cont'd)

- 2.1 <u>Undertaking of the Company</u> (Cont'd)
 - 2.1.3 <u>Liability of the Company</u> (Cont'd)
 - 2.1.3.14 The company shall not be liable for any act or omission concerning the implementation of presubscription, as defined herein.
 - 2.1.3.15 With respect to Telecommunications Relay Service (TRS), any service provided by Company which involves receiving, translating transmitting or delivering messages by telephone, text telephone, a telecommunications device for the deaf, or any other instrument over the facilities of Company or any connecting Carrier, Company's liability for the interruption or failure of the service shall not exceed an amount equal to the Company's charge for a one minute call to the called station at the time the affected call was made.
 - 2.1.3.16 The included tariff language does not constitute a determination by the Commission that a limitation of liability imposed by the Company should be upheld in a court of law. Acceptance for filing by the Commission recognizes that it is a court's responsibility to adjudicate negligence and consequential damage claims. It is also the court's responsibility to determine the validity of the exculpatory clause.
 - 2.1.4 <u>Notification of Service-Affecting Activities</u>

The Company will provide the Customer reasonable notification of service-affecting activities that may occur in normal operation of its business such activities may include, but are not limited to, equipment or facilities additions, removals or rearrangements and routine preventative maintenance. Generally, such activities are not specific to an individual Customer but affect many Customer's services. No specific advance notification period is applicable to all service activities. The Company will work cooperatively with the Customer to determine the reasonable notification requirements. With some emergency or unplanned service-affecting conditions, such as an outage resulting from cable damage, notification to the Customer may not be possible.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005

2. <u>REGULATIONS</u> (Cont'd)

Issued: March 3, 2005

2.1 <u>Undertaking of the Company</u> (Cont'd)

2.1.5 Provisions of Equipment and Facilities

- 2.1.5.1 The Company shall use reasonable efforts to make available services to a Customer on or before a particular date, subject to the provisions of and compliance by the Customer with, the regulations contained in this Price List. The Company does not guarantee availability by any such date and shall not be liable for any delays in commencing service to any Customer.
- 2.1.5.2 The Company shall use reasonable efforts to maintain facilities and equipment that it furnishes to the Customer. The Customer may not, nor may Customer permit others to, rearrange, disconnect, remove, attempt to repair, or otherwise interfere with any of the facilities or equipment installed by the Company, except upon the written consent of the Company.
- 2.1.5.3 The Company may substitute, change or rearrange any equipment or facility at any time and from time to time, but shall not thereby alter the technical parameters of the service provided the Customer.
- 2.1.5.4 Equipment the Company provides or installs at the Customer Premises for use in connections with the services the Company offers shall not be used for any purpose other than that for which the Company provided it.
- 2.1.5.5 The Customer shall be responsible for the payment of service charges as set forth herein for visits by the Company's agents or employees to the Premises of the Customer when the service difficulty or trouble report results from the use of equipment or facilities provided by any party other than the Company, including but not limited to the Customer.
- 2.1.5.6 The Company shall not be responsible for the installation, operation, or maintenance of any Customer provided communications equipment. Where such equipment is connected to the facilities furnished pursuant to this Price List, the responsibility of the Company shall be limited to the furnishing of facilities offered under this Price List and to the maintenance and operation of such facilities. Subject to this responsibility, the Company shall not be responsible for:
 - (a) the transmission of signals by Customer provided equipment or for the quality of, or defects in, such transmission; or
 - (b) the reception of signals by Customer provided equipment.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Alaine Miller, VP - Regulatory & External Affairs 1633 Westlake Avenue, No., Suite 200 Seattle, WA 98109

2. REGULATIONS (Cont'd)

2.1 <u>Undertaking of the Company</u> (Cont'd)

2.1.5 Provisions of Equipment and Facilities (Cont'd)

2.1.5.7 The company reserves the reasonable right to assign, designate or change telephone numbers, any other call number designations associated with access services, or the company serving central office prefixes associated with such numbers, when necessary in the conduct of its business.

2.1.6 Non-Routine Installation

At the Customer's request, installation and/or maintenance may be performed outside the Company's regular business hours or in hazardous locations. In such cases, charges based on cost of the actual labor, material, or other-costs incurred by or charged to the Company will apply. If installation is started during regular business hours but, at the Customer's request, extends beyond regular business hours into time periods including, but not limited to weekends, holidays, and/or night hours, additional charges may apply.

2.1.7 Special Construction

Subject to the arrangement of the Company and to all of the regulations contained in this Price List, special construction of facilities may be undertaken on a reasonable efforts basis at the request of the Customer. Special construction is that construction undertaken.

- (a) where facilities are not presently available, and there is not other requirement for the facilities so constructed;
- (b) of a type other than that which the Company would normally utilize in the furnishing of its services;
- (c) over a route other than that which the Company would normally utilize in the furnishing of its services;
- (d) in a quantity greater than that which the Company would normally construct;
- (e) on an expedited basis;
- on a temporary basis until permanent facilities are available; involving abnormal costs; or in advance of its normal construction.

Special construction charges will be determined as described on 4.1 following.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

2. REGULATIONS (Cont'd)

2.1 <u>Undertaking of the Company</u> (Cont'd)

2.1.8 Ownership of Facilities

Title to all facilities provided in accordance with Price List remains with the Company, its agents or contractors.

2.2 <u>Prohibited Uses</u>

- 2.2.1 The services the Company offers shall not be used for any unlawful purpose or for any use as to which the Customer has not obtained all required governmental approvals, authorizations, license, consents and permits.
- 2.2.2 The Company may require applicants for service who intend to use the company offerings for resale and/or for shared use to file a letter with the company confirming that their use of the Company's offerings complies with relevant laws and PUC regulations, policies, orders, and decisions.
- 2.2.3 The Company may require a Customer to immediately shut down its transmission of signals if said transmission is causing interference to others.

2.3 Obligations to the Customer

- 2.3.1 The Customer shall be responsible for:
 - (a) the payment of all applicable charges pursuant to this Price List;
 - (b) damage to or loss of Company facilities or equipment caused by the acts or omissions of the Customer; or the noncompliance by the Customer, with these regulations; or by fire or theft or other casualty on the Customer Premises, unless caused by the negligence or willful misconduct of the employees or agents of the Company;

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

2. REGULATIONS (Cont'd)

2.3 Obligations to the Customer (Cont'd)

- 2.3.1 The Customer shall be responsible for:
 - providing at no charge, as specified from time to time by the company, any (c) needed personnel, equipment, space and power to operate Company facilities and equipment installed on the premises of the Customer Premises, and the level of heating and air conditioning necessary to maintain the proper operating environment on such premises;
 - (d) obtaining, maintaining, and otherwise having full responsibility for all rightsof-way and conduit necessary for installation of fiber optic able and associated equipment used to provide Network Services to the Customer from the Company's designated point of termination or property line to the location of the equipment space described in 2.3.1 (c). Any costs associated with the obtaining and maintaining the rights-of-way described herein, including the costs of altering the structure to permit installation of Company provided facilities, shall be bored entirely by, or may be charged by the Company to, the Customer;
 - (e) providing a safe place to work and complying with all laws and regulations regarding the working conditions on the premises at which Company employees and agents shall be installing or maintaining Company facilities and equipment. The Customer may be required to install and maintain Company facilities and equipment within a hazardous area if, in the Company opinion, injury or damage to Company employees or property might result from installation or maintenance by the Company. The Customer shall be responsible for identifying, monitoring, removing and disposing of any hazardous material (e.g. friable asbestos) prior to any construction or installation work;
 - (f) complying with all laws and regulations applicable to, and obtaining all consents, approvals, licenses and permits as may be required with respect to the location of Company facilities and equipment in any Customer premises or the right-of-way for which Customer is responsible under section 2.3.1 (d); and granting or obtaining permission for Company agents or employees to enter the premises of the Customer at any time for the purpose of installing, inspecting, maintaining, repairing, or upon termination of service as stated herein, removing the facilities or equipment of the Company; and
 - not creating or allowing to be placed any liens or other encumbrances on (g) Company equipment or facilities.

Idaho Public Utilities Commission Office of the Secretary ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Effective: March 13, 2005

Issued: March 3, 2005

Alaine Miller, VP - Regulatory & External Affairs 1633 Westlake Avenue, No., Suite 200

2. <u>REGULATIONS</u> (Cont'd)

2.3 Obligations to the Customer (Cont'd)

2.3.2 Claims

With respect to any service or facility provided by the Company, customer shall indemnify, defend and hold harmless the Company from and against all claims, actions, damages, liabilities, costs and expenses, including reasonable attorney fees for:

- (a) any loss, destruction or damage to property of the Company or any third party, or the death or injury to persons, including, but not limited to, employees or invitees of either party, to the extent caused by or refuting from the negligent or intentional act or omission of the Customer, its employees, agents, representatives or invitees; or
- (b) any claim, loss, damage, expense or liability for infringement of any copyright, patent, trade secret, or any proprietary or intellectual property right of any third party, arising from any act or omission by the Customer, including, without limitation, use of Company services and facilities in a manner not contemplated by the agreement between Customer and the Company.

2.3.3 <u>Jurisdictional Reporting</u>

The jurisdictional reporting requirements will be as specified below. When a Customer orders Access Service, its projected Percent Interstate Usage (PIU) must be provided in whole numbers to the Company. These whole number percentages will be used by the Company to apportion the use and/or charges between interstate and intrastate until a revised report is received as set forth herein.

- 2.3.3.1 Originating Access: Originating access minutes may be based on traffic originating at the State, LATA or local Switching Center level, provided that the traffic being measured is only traffic originating from the Company Local Switching Center(s). The Customer must provide the Company with a projected PIU factor on a quarterly basis, as specified below. Originating access minutes will be measured as follows, based on type of access:
 - 2.3.3.1.1 For Feature Group D Switched Access Services(s), as defined in Section 5.2.1, where the Company can determine jurisdiction by its call detail, the projected Percent Interstate Usage (PIU) will be developed by the Company on a monthly basis by dividing the measured interstate originating access minutes by the total originating access minutes.

Office of the Secretary

ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

(D)

INTRASTATE ACCESS SERVICES PRICE LIST

2. <u>REGULATIONS</u> (Cont'd)

- 2.3 Obligations to the Customer (Cont'd)
 - 2.3.3 Jurisdictional Reporting (Cont'd)
 - 2.3.3.1.2 For Feature Group D with 950 Access, as defined in Section 5.5.3.1, the Customer must provide the Company with a projected PIU factor by supplying the Company with an interstate percentage of originating access minutes.
 - 2.3.3.1.3 For 500, 700, 800 and operator service access, the Customer must provide the Company with a projected PIU factor for each type of access. The Customer who provides a PIU factor shall supply the Company with an interstate percentage of originating access minutes.
 - 2.3.3.1.4 If no PIU for originating minutes is submitted as specified herein, then the projected PIU will be set on a default basis of 50 percent interstate traffic and 50 percent intrastate traffic.
 - 2.3.3.2 <u>Terminating Access</u>: For Feature Group D Switched Access Service(s), the Customer must provide the company with a projected PIU factor by supplying the Company with an interstate percentage of terminating access minutes on a quarterly basis, as described in Section 2.3.3.4 below. If no projected PIU factor is submitted by the Customer, then the projected PIU will be set on a default basis of 50 percent interstate traffic and 50 percent intrastate traffic.
 - 2.3.3.3 Except where the Company measured access minutes are used as set forth in 2.3.3.1 above, the Customer reported projected PIU factor as set forth above will be used until the Customer reports a different projected PIU factor, as set forth below. The revised report will serve as the basis for future billing and will be effective on the next bill date.
 - 2.3.3.4 Effective on the first of January, April, July and October of each year the Customer shall update its interstate and intrastate jurisdictional report. The Customer shall forward to the Company, to be received no later than 15 days after the first of such month, a revised report showing the interstate and intrastate percentage of use for the past three months ending the last day of December, March, June and September, respectively, for each service arranged for interstate use, based solely on the traffic originating from or terminating to the Company Local Switching Center. The revised report will serve as the basis for the next three months billing and will be effective on the bill date for that service. If the Customer does not supply the reports for those services where reports are

Effective: November 13, 2011

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

Kelly Faul – Regulatory Affairs Director

Issued: November 3, 2011

2. <u>REGULATIONS</u> (Cont'd)

- 2.3 <u>Obligations to the Customer</u> (Cont'd)
 - 2.3.3 <u>Jurisdictional Reporting (Cont'd)</u>

needed, the Company will assume the percentage to be the same as that provided previously. For those cases in which a quarterly report has never been received from the Customer, the Company will assume the percentages to be the same as those provided in the Access Service Request.

2.3.3.5 <u>Jurisdictional Reports Verification</u>: For Switched Access Service, if a billing dispute arises or a regulatory commission questions the projected PIU factor, the customer will provide the data issued to determine the projected PIU factor. The Customer will supply the data within 30 days of the Company request.

The Customer shall keep records of call detail from which the percentage of interstate and intrastate use can be ascertained and, upon request of the Company, shall make the records available for inspection as reasonably necessary for purposes of verification of the percentages. The Company reserves the right to conduct an audit at any time during the year. The Customer, at its own expense, has the right to retain an independent auditing firm.

For Special Access Service*, if a billing dispute arises or a regulatory commission questions the projected PIU factor, the Company will ask the Customer to provide the data the Customer used to determine the certified interstate percentage. The Customer shall supply the data within 30 days of the Company request. The Customer shall keep records of system design and functions from which the percentage was determined, and, upon request of the Company, shall make the records available for inspection as reasonably necessary for purposes of verifications of the percentages.

2.3.3.6 <u>Determination of Jurisdiction of Mixed Use Special Access Service*</u>:

When an ASR is submitted for interstate and intrastate Special Access Service or the Customer uses such services, the Customer will provide to the Company an estimate of whether the interstate traffic will comprise more than 10 percent or less than 10 percent of total traffic. For existing services, the Customer is required to certify the jurisdiction of its service.

(a) If the Customer estimates that the interstate traffic on the service involved constitutes 10 percent or less of the Customer's total traffic on that service, the service will be provided in accordance with the applicable rules and regulations of this Price List.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

Effective: May 8, 2020

Kelly Faul – Senior Manager 22001 Loudoun County Pkwy Ashburn, VA 20147

Issued: April 21, 2020

(C)

(C)

(N)

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING
May 8, 2020
Boise, Idaho

- 2. <u>REGULATIONS</u> (Cont'd)
 - 2.3 Obligations to the Customer (Cont'd)
 - 2.3.3 <u>Jurisdictional Reporting</u> (Cont'd)
 - (b) If the Customer estimates that the interstate traffic on the service involved constitutes more than 10 percent of the Customer's total traffic on that service, the service will be provided in accordance with the applicable rules and regulations of XO Communications Services, Inc.'s Price List, F.C.C. No. 1

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

FEB 3 - 2012

(Some material previously located on this page is now on page 22.6.)

Boise, Idaho

Issued: January 24, 2012 Effective: February 3, 2012

2. <u>REGULATIONS</u> (Cont'd)

2.3 Obligations to the Customer (Cont'd)

2.3.4 <u>Identification and Rating of Toll VoIP-PSTN Traffic</u>

A. Scope

- 1. Toll VoIP-PSTN Traffic is a Customer's interexchange voice traffic that (1) is exchanged with the Company in Time Division Multiplexing ("TDM") format over Public Switched Telephone Network ("PSTN") facilities and (2) originates and/or terminates in Internet Protocol ("IP") format. Toll VoIP-PSTN Traffic originates and/or terminates in IP format when it originates from and/or terminates to an end user of a service that requires IP-compatible customer premises equipment.
- 2. In the absence of an agreement between the Company and the Customer specifying the treatment of Toll VoIP-PSTN Traffic, this section governs the treatment of Toll VoIP-PSTN Traffic pursuant to 47 C.F.R. § 51.913 and the Federal Communications Commission's orders adopting and amending those rules ("Section 51.913").
- 3. This section will be applied to identify intrastate Toll VoIP-PSTN Traffic exchanged with a Customer or its affiliate that is a local exchange carrier only to the extent that the Customer or its local exchange carrier affiliate has also implemented billing for intrastate Toll VoIP-PSTN Traffic in accordance with Section 51.913.

B. Rating of Toll VoIP-PSTN Traffic

All interstate Toll VoIP-PSTN Traffic and intrastate terminating Toll VoIP-PSTN Traffic will be billed at rates equal to the Company's relevant interstate switched access rates as provided in section 6 of Tariff FCC No. 1 at http://www.xo.com/legal-and-privacy/public-policy/xo-state-tariffs/.

Any intrastate originating Toll VoIP-PSTN Traffic identified in accordance with this tariff section will be billed at rates equal to the Company's relevant interstate originating switched access rates.

(T) (T)

C. <u>Calculation and Application of Percent-VoIP-Usage Factor</u>

The Company will determine the number of intrastate originating Toll VoIP-PSTN Traffic minutes of use ("MOU") to which the relevant interstate originating switched access rates will be applied under section 2.3.4.B, above, by applying a Percentage VoIP Usage ("PVU") factor to the total intrastate originating switched access MOU sent by the Company to the Customer. The PVU will be derived and applied as follows:

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING
OCT 1 3 2014

301 2 0 =-

Boise, Idaho

Issued: October 3, 2014 Effective: October 13, 2014

(C)

INTRASTATE ACCESS SERVICES PRICE LIST

2. REGULATIONS (Cont'd)

2.3 Obligations to the Customer (Cont'd)

2.3.4 <u>Identification and Rating of Toll VoIP-PSTN Traffic (Cont'd)</u>

C. Calculation and Application of Percent-VoIP-Usage Factor (Cont'd)

- 1. The Customer will calculate and furnish to the Company a factor (the "PVU-C") representing the percentage of the total intrastate originating switched access MOU that the Customer receives from the Company in the state and originates in IP format. This PVU-C shall be based on information such as the number of the Customer's retail VoIP subscriptions in the state (e.g., as reported on FCC Form 477), traffic studies, actual call detail, or other relevant and verifiable information.
- 2. The Company will, likewise, calculate a factor (the "PVU-X") representing the percentage of the Company's total intrastate originating switched access MOU in the state that the Company originates on its network in IP format. This PVU-X shall be based on information, such as the number of the Company's retail VoIP subscriptions in the state (e.g., as reported on FCC Form 477), traffic studies, actual call detail, or other relevant and verifiable information.
- 3. The Company will use the PVU-C and PVU-X factors to calculate a PVU factor that represents the percentage of total intrastate originating switched access MOU sent by the Company to the Customer that is originated or terminated in IP format, whether at the Company's end, at the Customer's end, or at both ends. The PVU factor will be calculated as the sum of: (A) the PVU-C factor and (B) the PVU-X factor times (1.0 minus the PVU-C factor).
- The Company will apply the PVU factor to the total intrastate originating switched access MOU
 received from the Customer in order to determine the number of intrastate originating Toll VoIPPSTN Traffic MOU.
 - Example 1: The PVU-C is 40% and the PVU-X is 10%. The PVU factor is equal to $40\% + (10\% \times 60\%) = 46\%$. The Company will bill 46% of the Customer's intrastate originating switched access MOU at the Company's tariffed interstate originating switched access rates.
 - Example 2: The PVU-C is 0% and the PVU-X is 10%. The PVU factor is equal to $0\% + (10\% \times 100\%) = 10\%$. The Company will bill 10% of the Customer's intrastate originating switched access MOU at the Company's tariffed interstate originating switched access rates.
 - Example 3: The PVU-C is 100%. No matter what the PVU-X factor is, the PVU is 100%. The Company will bill 100% of the Customer's intrastate originating switched access MOU at the Company's tariffed interstate originating switched access rates.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

AUG 1 6 2014

Boise, Idaho

Issued: August 6, 2014 Effective: August 16, 2014

(C)

(C)

(C)

INTRASTATE ACCESS SERVICES PRICE LIST

REGULATIONS (Cont'd)

2.3 Obligations to the Customer (Cont'd)

2.3.4 Identification and Rating of Toll VoIP-PSTN Traffic (Cont'd)

C. Calculation and Application of Percent-VoIP-Usage Factor (Cont'd)

- 5. The Customer shall not modify its reported PIU factor to account for Toll VoIP-PSTN Traffic.
- 6. The Customer shall retain any call detail, work papers and information used to develop the PVU-C factor for a minimum of one year.
- 7. If the Customer does not furnish the Company with a PVU-C pursuant to section 2.3.4.C.1, above, the Company will utilize a PVU-C equal to zero.

D. Initial PVU factor

- 1. If the PVU factor cannot be implemented in the Company's billing systems by July 1, 2014, once the factor can be implemented, the Company will adjust the Customer's bills to reflect the PVU factor retroactively to July 1, 2014. In calculating the initial PVU, the Company will take the Customer-specified PVU-C into account retroactively to July 1, 2014, provided that the Customer provides the PVU-C factor to the Company no later than October 15, 2014; otherwise, the Company will set the initial PVU-C equal to zero and calculate the PVU accordingly.
- The Company may choose to provide credits based on the PVU-C and PVU-X factors on a quarterly basis until such time as billing system modifications can be implemented.

E. PVU Factor Updates

- 1. The Customer may update the PVU-C factor quarterly using the method set forth in section 2.3.4.C.1, above. If the Customer chooses to submit such updates, it shall provide to the Company, no later than 15 days after the first day of January, April, July and/or October of each year, a revised PVU-C factor based on data for the prior three months, ending the last day of December, March, June, and September, respectively.
- The Company may, likewise, update the PVU-X factor quarterly based on data for the prior three months.
- 3. The Company will use any revised PVU-C and/or PVU-X factors to calculate a revised PVU that will apply prospectively and serve as the basis for subsequent billing until superseded by a new PVU. No prorating or backbilling will be done based on the updated PVU-C or PVU-X factors.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

AUG 1 6 2014

Boise, Idaho

Issued: August 6, 2014 Effective: August 16, 2014

2. <u>REGULATIONS</u> (Cont'd)

2.3 Obligations to the Customer (Cont'd)

2.3.4 Identification and Rating of Toll VoIP-PSTN Traffic (Cont'd)

F. PVU Factor Verification

Not more than twice in any year, the Company may request from the Customer an
overview of the process used to determine the PVU-C factor, the call detail records,
description of the method for determining how the end user originates or terminates
calls in IP format, and other information used to determine the Customer's PVU-C
factors furnished to the Company in order to validate the PVU-C factor supplied. The
Customer shall supply the requested data and information within 30 days of the
Company's request.

(C)

- 2. The Company may dispute the Customer's PVU-C factor based upon:
 - (a) A review of the requested data and information provided by the Customer.
 - (b) The Company's review of other market information, FCC reports on VoIP lines, such as FCC Form 477 or state level results based on FCC Local Competition Report or other relevant data.
 - (c) A change in the reported PVU-C factor by more than five percentage points from the preceding quarter.
- 3. If, after review of the data and information, the Customer and the Company agree upon a revised PVU-C factor, the Customer and the Company will begin using that revised PVU-C factor with the next bill period.
- 4. If the dispute is unresolved, the Company may initiate an audit. The Company shall limit audits of the Customer's PVU-C factor to no more than two times per year. The Customer may request that the audit be conducted by an independent auditor. In such cases, the associated auditing expenses will be paid by the Customer.
 - (a) In the event that the Customer fails to provide adequate records to enable the Company or an independent auditor to conduct an audit verifying the Customer's PVU-C factor, the Company will bill the usage for all contested periods using the most recent undisputed PVU-C factors reported by the Customer. This PVU-C factor will remain in effect until the audit can be completed.
 - (b) The Company will adjust the Customer's PVU-C factor based on the results of the audit and implement the revised PVU-C in the next billing period or quarterly report date, whichever is first. The revised PVU-C factor will apply for the next two quarters before new factors can be submitted by the Customer.
 - (c) If the audit supports the disputed PVU-C factor submitted by the Customer, the usage for the contested periods will be adjusted to reflect such PVU-C factor.

Idaho Public Uttlities Commencerive: February 26, 2012
Office of the Secretary

Kelly Faul – Regulatory Affairs Director

ACCEPTED FOR FILING

FEB 2 6 2012

2. REGULATIONS (Cont'd)

2.3 Obligations to the Customer (Cont'd)

2.3.4 <u>Call Signaling</u>

Depending on the signaling system used by the Customer in its network, the Customer's facilities shall transmit the following call signaling information, consistent with the requirements of 47 C.F.R. § 64.1600 et seq, as may be amended by the Federal Communications Commission, to the Company on traffic that the Customer delivers for termination on the Company's network.

(C)

A. Signaling System 7 (SS7) Signaling

When the Customer uses SS7 signaling, it will transmit the Calling Party Number (CPN) or, if different from the CPN, the Charge Number (CN) information in the SS7 signaling stream.

B. Multi-Frequency (MF) Signaling

When the Customer uses MF signaling, it will transmit the number of the calling party or, if different from the number of the calling party, the Charge Number (CN) information in the MF Automatic Number Identification (ANI) field.

C. Internet Protocol (IP) Signaling

When the Customer uses IP signaling, it will transmit the telephone number of the calling party or, if different from the telephone number, the billing number of the calling party.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

JUL 1 3 2012

Effective: July 13, 2012

Issued: July 2, 2012

(M)

INTRASTATE ACCESS SERVICES PRICE LIST

2. REGULATIONS (Cont'd)

2.4 <u>Customer Equipment and Channels</u>

2.4.1 In General

A Customer may transmit or receive information or signals via the facilities of the Company.

2.4.2 Station Equipment

Facilities and equipment to Company-owned facilities and equipment.

- 2.4.2.1 Customer provided terminal equipment on the Customer Premises, and the electric power consumed by such equipment shall be provided and maintained at the expense of the Customer.
- 2.4.2.2 The Customer is responsible for ensuring that Customer-provided equipment connected to Company equipment and facilities is compatible with such equipment and facilities. The magnitude and character of the voltages and currents impressed on Company-provided equipment and wiring by the connection, operation, or maintenance of such equipment and wiring shall be such as not to cause damage to the Company-provided equipment and wiring or injury to the Company's employees or to other persons. Any additional protective equipment required to prevent such damage or injury shall be provided by the Company at the Customer's expense.

2.4.3 Interconnection of Facilities

- 2.4.3.1 Any special interface equipment necessary to achieve compatibility between the facilities and equipment of the Company used for furnishing Network Services and the channels, facilities, or equipment of others, including Channel Service Units ("CSU") shall be provided at the Customer's expenses.
- 2.4.3.2 Access Services may be connected to the services or facilities of other communications companies only when authorized by, and in accordance with, the terms and conditions of the Price Lists of the other communications companies which are applicable to such connections.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

(M)

FEB 3 - 2012

Boise, Idaho

Issued: January 24, 2012 Effective: February 3, 2012

2. <u>REGULATIONS</u> (Cont'd)

2.4 <u>Customer Equipment and Channels</u> (Cont'd)

2.4.4 <u>Inspection and Testing</u>

- 2.4.4.1 Upon suitable notification to the Customer, and at a reasonable time, the Company may make such tests and inspections as may be necessary to determine that the Customer is complying with the requirements set forth in section 2.4.2.2 for the installation, operation, and maintenance of Customer-provided facilities, equipment, and wiring in the connection of Customer-provided equipment.
- 2.4.4.2 If the protective requirements for Customer provided equipment are not being complied with, the Company may take such action as it deems necessary to protect its facilities, equipment, and personnel. The Company will notify the Customer promptly if there is any need for further corrective action and notify the Company of the action taken, if the customer fails to do this, the Company may take whatever additional action is deemed necessary, including the suspension of service, to protect its facilities, equipment and personnel from harm

2.5 Payment Arrangements

2.5.1 Payment for Service

The Customer is responsible for the payment of all charges for facilities and services furnished by the Company to the Customer.

2.5.1.1 Taxes

The Customer is responsible for the payment of any sales, use, gross receipts, excise, access or other local, state and federal taxes, charges or surcharges (however designated) (excluding taxes on the Company's net income) imposed on or based upon the provision, sale or use of Access Services. All such taxes shall be separately designated on the Company's invoices. The Company will charge for the Idaho gross receipts tax on the Company's invoice for Local Access Services. Any taxes imposed by local jurisdiction (e.g., county and municipal taxes) will only be recovered from those Customers located in the affected jurisdictions. If an entity other than the company (e.g., another carrier or a supplier) imposes charges on the Company, in addition to its own internal costs, in connection with a service for which the Company's Non-recurring Charge is specified, those charges will be passed on to the Customer. It shall be the responsibility of the Customer to pay any such taxes that subsequently become applicable retroactively.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005

2. <u>REGULATIONS</u> (Cont'd)

2.5 <u>Payment Arrangements (Cont'd)</u>

2.5.1 Payment for Service (Cont'd)

2.5.1.2 A surcharge is imposed on all charges for service originating at addresses in states which levy, or assert a claim of right to levy, a gross receipts tax on the Company's operations in any such state, or a tax on interstate access charges incurred by the Company for originating access to telephone exchanges in that state¹. This surcharge is based on the particular state's receipts tax and other state taxes imposed directly or indirectly upon the Company by virtue of, and measured by, the gross receipts or revenues of the Company in that state and/or payment of interstate access charges in that state. The surcharge will be shown as a separate line item on the Customer's monthly invoice.

2.5.2 <u>Billing and Collection of Charges</u>

The Customer is responsible for payment of all charges incurred by the Customer or other users for services and facilities furnished to the Customer by the Company.

The Company shall bill on a current basis all charges incurred by, and credits due to, the Customer under this Price List attributable to services established, provided, or discontinued during the preceding business cycle.

- 2.5.2.1 Non-recurring charges are due and payable within 30 days after the date an invoice is mailed to the Customer by the Company.
- 2.5.2.2 The Company shall present invoices for non-usage sensitive Recurring Charges monthly to the Customer, in advance of the month in which service is to be provided, and invoices for usage sensitive charges monthly to the Customer subsequent to the usage. Recurring and usage sensitive charges shall be due and payable within 30 days after the invoice date.
- 2.5.2.3 When service does not begin on the first day of the month, or end on the last day of the month, the charge for the action of the month in which service was furnished will be calculated on a pro rata basis. For this purpose, every month is considered to have 30 days.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Issued: March 3, 2005

Boise, Idaho

Effective: March 13, 2005

.......

Pending the conclusion of any challenge to the jurisdiction's right to impose a gross receipts tax the Company may elect to impose and collect a surcharge covering such taxes, unless otherwise constrained by court order or direction, or it may elect not to impose and collect the surcharge and the challenged tax is found to have been invalid and unenforceable, the Company will credit or refund such amounts to affected Customers (less its reasonable administrative costs), if the funds collected were retained by the Company or if they were delivered over to the taxing jurisdiction and later returned to the Company, or negotiate an arrangement with the taxing jurisdiction that benefits Customers in the jurisdiction in the future.

2. REGULATIONS (Cont'd)

- 2.5 Payment Arrangements (Cont'd)
 - 2.5.2 <u>Billing and Collection of Charges</u> (Cont'd)
 - 2.5.2.4 Billing of the Customer by the Company will begin on the Service Commencement Date, which is the first day following the date on which the company notifies the Customer that the service or facility is available for use, except that the Service Commencement Date may be postponed by mutual agreement of the parties, or if the service or facility does not conform to standards set forth in this Price List or the Service order. In any case, the Service Commencement Date shall be no later than the first day that Customer begins using the Service. Billing accrues through and includes the day that the service, circuit, arrangement or component is discontinued.
 - 2.5.2.5 If any portion of the payment is received by the Company after the date due, or if any portion of the payment is received by the Company in funds which are not immediately available, then a late payment penalty shall be due to the Company. The late payment penalty shall be a portion of the payment not received by the due date, multiplied by a late factor of 1.5% which is compounded monthly.
 - 2.5.2.6 Ordering, Rating and Billing of Access Services Where More Than One Exchange Carrier is Involved

All Recurring and Non-Recurring Charges for services provided by each Exchange Carrier are billed under each Company's applicable Price Lists. Under a Meet Point Billing arrangement, the Company will bill for charges for traffic carried between the Company Local Switching Center and the End User and for the portion of any transport facilities provided by the Company between the Customer's location and the Company's local switching center.

The multiple billing arrangement described in this section is subject to the provisions of the Multiple Exchange Carrier Access Billing Guidelines (MECAB) and the Multiple Exchange Carrier Ordering and Design Guidelines (MECOD).

The Company must notify the Customer of: 1) the meet point option that will be used; 2) the Carrier(s) that will render the bill(s); 3) the Carrier(s) to whom payment should be remitted; and 4) the Carrier(s) that will provide the bill inquiry function. The Company shall provide such notification at the time orders are placed for Access Service. Additionally, the Company shall provide this notice in writing 15 days in advance of any changes in the arrangement.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

2. <u>REGULATIONS</u> (Cont'd)

2.5 <u>Payment Arrangements</u> (Cont'd)

2.5.2 <u>Billing and Collection of Charges</u> (Cont'd)

The Company will handle the ordering, rating and billing of Access Services under this Price List where more than one Exchange Carrier is involved in the provision of Access Services, as follows:

- (1) The Company must receive an order for Feature Group D (FGD) Switched Access Service, as defined herein, ordered to the Company's Local Switching Center through a switch operated by another Exchange Carrier.
- (2) In addition, for FGD Switched Access Service ordered to the Company's Local Switching Center through a switch operated by another Exchange Carrier with whom the Company has an agreement, the Customer may be required to submit an order as specified by the Exchange Carrier which operates the switch.
- (3) Separate bills will be rendered by the Exchange Carrier for FGD access service.
- (4) Rating and Billing of Service: Each Company will provide its portion of access service based on the regulations, rates and charges contained in its respective Access Service Price List, subject to the following rules, as appropriate:
 - (a) The application of non-distance sensitive rate elements varies according to the rate structure and the location of the facilities involved:
 - (i) when rates and charges are listed on a per minute basis, the Company's rates and charges will apply to traffic originating from the Customer's Premises and terminating at the End User's Premises, and vice versa.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

2. <u>REGULATIONS</u> (Cont'd)

2.5 Payment Arrangements (Cont'd)

2.5.3 Billing or Payment Disputes

2.5.3.1 Customer Must Dispute Bills or Rates Charged Within 30 Days of Invoice

If a Customer does not give the Company written notice of a billing or rate dispute within 30 days from the date the invoice was rendered, such invoice and the charges levied shall be deemed to be reasonable, correct and binding. In the event Customer disputes any billing or rates, Customer shall notify Company in writing, providing the billing identification, trouble ticket number and an explanation for the dispute, and shall nevertheless pay all undisputed charges within the 30 day period specified above.

2.5.3.2 Dispute Procedures

The date of the dispute shall be the date on which the Customer furnishes the Company the account information required above.

Company will investigate the dispute and attempt to resolve the billing or rate issues within 45 days. Payment shall not prejudice Customer's right to dispute charges, so long as they are disputed in the manner and within the 30 days specified in this Section. No action or proceeding against the Company disputing bills or rates charged shall be commenced unless Customer has first complied with this Section, or in any event more than 90 days after the service is rendered.

The Date of Resolution is the date the Company completes its investigation, provides written notice to the Customer regarding the disposition of the claim, i.e., resolved in favor of the Customer or resolved in favor of the Company, and credits the Customer's account, if applicable.

Upon resolution of dispute, Customer must make payment to Company within 15 days from the Date of Resolution.

In the event that a billing dispute concerning any charges billed to the Customer by the Company is resolved in favor of the Company, any payments withheld pending settlement of the dispute shall be subject to the late payment penalty set forth in 2.5.2.5. If the Customer pays the bill in full by the payment due date, but within 90 days initiates a billing dispute and withholds the disputed amount from a subsequent bill past the due date, penalty interest may be applicable.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

NOV 2 2 2009

Boise, Idaho

(C)(T)

(C)(T)

Issued: November 12, 2009

Effective: November 22, 2009

2. REGULATIONS (Cont'd)

2.5 Payment Arrangements (Cont'd)

2.5.3 Billing or Payment Disputes (Cont'd)

2.5.3.2 <u>Dispute Procedures (Cont'd)</u>

If the dispute is resolved in favor of the Company and the Customer has paid the disputed amount on or before the payment due date, no interest credit or penalties will apply.

If the Customer pays the bill in full by the payment due date, but after 90 days initiates a billing dispute and withholds the disputed amount from a subsequent bill past the due date, penalty interest may be applicable.

If the billing dispute is resolved in favor of the Customer, the Customer shall receive a credit from the Company. This credit will be an amount equal to the disputed amount resolved in the Customer's favor.

Adjustments for the quantities of services established or discontinued in any billing period beyond the minimum period set forth for services in other sections of this tariff will be prorated to the number of days based on a 30-day month.

(C)(T) (D) (D)

(C)(T)

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

NOV 2 2 2009

Boise, Idaho

Issued: November 12, 2009

Effective: November 22, 2009

2. <u>REGULATIONS</u> (Cont'd)

2.5 Payment Arrangements (Cont'd)

2.5.4 Advance Payments

To safeguard its interests, the Company may require a Customer to make an advance payment before services and facilities are furnished. The advance payment will not exceed an amount equal to the nonrecurring charge(s) and one month's charges for the service or facility. In addition, where special construction is involved, the advance payment may also include an amount equal to the estimated nonrecurring charges for the special construction and recurring charges (if any) for a period to be set between the Company and the Customer. The advance payment will be credited to the Customer's initial bill. An advance payment may be required in addition to a deposit.

2.5.5 Deposits

- 2.5.5.1 To safeguard its interest, the Company may require a Customer to make a deposit to be held as a guarantee for the payment of charges. A deposit does not relieve the Customer of the responsibility for the prompt payment of bills on presentation. The deposit will not exceed an amount equal to:
 - (a) two month's charges for a service or facility which has a minimum payment period of one month; or
 - (b) the charges that would apply for the minimum payment period for a service or facility which has a minimum payment period of more than one month; except that the deposit may include an additional amount in the event that a termination charge is applicable. At the Company's option, such deposit may be refunded to the Customer's account at any time. Also, the Company reserves the right to cease accepting and processing Service Orders after it has requested a security deposit and prior to the Customer's compliance with this request.
- 2.5.5.2 A deposit may be required in addition to an advance payment.
- 2.5.5.3 When a service or facility is discontinued, the amount of a deposit, if any, will be applied to the Customer's account and any credit balance remaining will be refunded. Before the service or facility is discontinued, the Company may, at its option, return the deposit or credit it to the Customer's account.
- 2.5.5.4 Deposits held will accrue simple interest at the rate specified by the Idaho Public Service Commission.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005

2. <u>REGULATIONS</u> (Cont'd)

2.5 Payment Arrangements (Cont'd)

2.5.6 <u>Discontinuance of Service</u>

- 2.5.6.1 Upon nonpayment of any amount owing the Company, the Company may, by giving ten day's prior written notice to the Customer, discontinue or suspend service without incurring any liability.
- 2.5.6.2 Upon violation of any of the other material terms or conditions for furnishing service the Company may, by giving 30 days prior written notice to the Customer, discontinue or suspend service without incurring any liability if such violation continues during that period.
- 2.5.6.3 Upon condemnation of any material portion of the facilities used by the Company to provide service to a Customer or if a casualty renders all or any material portion of such facilities inoperable beyond feasible repair, the Company, by notice to the Customer, may discontinue or suspend service without incurring any liability.
- 2.5.6.4 Upon the Customer's insolvency, assignment for the benefit of creditors, filing for bankruptcy or reorganization, or failing to discharge an involuntary petition within the time permitted by law, the Company may immediately discontinue or suspend service without incurring any liability.
- 2.5.6.5 Upon any governmental prohibition or required alteration of the services to be provided or any violation of an applicable law or regulation, the Company may immediately discontinue service without incurring any liability.
- 2.5.6.6 Upon the Company's discontinuance of service to the Customer under section 2.5.6.1 or 2.5.6.2, the Company, in addition to all other remedies that may be available to the Company at law or in equity or under any other provision of this Price List, may declare all future monthly and other charges which would have been payable by the Customer during the remainder of the term for which such services would have otherwise been provided to the Customer to be immediately due and payable (discounted to present value at six percent).

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005 Effective: March 13, 2005

2. REGULATIONS (Cont'd)

- 2.5 Payment Arrangements (Cont'd)
 - 2.5.6 <u>Discontinuance of Service</u> (Cont'd)
 - 2.5.6.7 When Access Service is provided by more than one Company, the companies involved in providing the joint service may individually or collectively deny service to a Customer for non-payment. Where the Company(s) affected by the nonpayment is incapable of effecting discontinuance of service without cooperation from the other joint providers of Switched Access Service, such other Company(s) will, if technically feasible, assist in denying the joint service to the Customer. Service denial for such joint service will only include calls originating or terminating within, or transiting, the operating territory of the Company initiating the service denial for nonpayment. When more than one of the joint providers must deny service to effectuate termination for nonpayment, in cases where a conflict exists in the applicable Price List provisions, the Price List regulations of the company whose Local Switching Center serves the Customer shall apply for joint service discontinuance.
 - 2.5.6.8 The Company may discontinue the furnishing of any and/or all service(s) to a Customer, without incurring any liability:
 - 2.5.6.8.1 Immediately and without notice if the Company deems that such action is necessary to prevent or to protect against fraud or to otherwise protect its personnel, agents, facilities or services. The Company may discontinue service pursuant to this sub-section 2.5.6.8.1 a-f), if
 - (a) The Customer refuses to furnish information to the Company regarding the Customer's credit-worthiness, its past or current use of Common Carrier communications services or its planned use of service(s); or
 - (b) The Customer provides false information to the Company regarding the Customer's identity, address, credit-worthiness, past or current use of Common Carrier communications services, or its planned use of the Company's service(s); or
 - (c) The Customer states that it will not comply with a request of the Company for security for the payment for service(s) in accordance with Section 2.5.5.1 above; or

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

2. <u>REGULATIONS</u> (Cont'd)

- 2.5 <u>Payment Arrangements</u> (Cont'd)
 - 2.5.6 <u>Discontinuance of Service</u> (Cont'd)
 - (d) The Customer has been given written notice by the Company of any past due amount (which remains unpaid in whole or in part) for any of the Company's other Common Carrier communications services to which the Customer either subscribes or had subscribed or used; or
 - (e) The Customer uses service to transmit a message, locate a person or otherwise give or obtain information without payment for the service; or
 - (f) The Customer uses, or attempts to use, service with the intent to avoid the payment, either in whole or in part, of the Price Listed charges for the service by:
 - (1) Using or attempting to use service by rearranging, tampering with, or making connections to the Company's service not authorized by this Price List; or
 - Using tricks, schemes, false or invalid numbers, false credit devices, electronic devices; or
 - (3) Any other fraudulent means or devices.
 - 2.5.6.8.2 Immediately upon written notice to the Customer of any sum thirty (30) days past due;
 - 2.5.6.8.3 Immediately upon written notice to the Customer, after failure of the Customer to comply with a request made by the Company for security for the payment of service in accordance with Section 2.5.5.1, above; or
 - 2.5.6.8.4 Seven (7) days after sending the Customer written notice of noncompliance with any provision of this Price List if the noncompliance is not corrected within that seven (7) day period. The discontinuance of service(s) by the Company pursuant to this Section does not relieve the Customer of any obligation to pay the Company for charges due and owing for service(s) furnished up to the time of discontinuance.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005

2. REGULATIONS (Cont'd)

2.5 Payment Arrangements (Cont'd)

2.5.7 Cancellation of Application for Service

- 2.5.7.1 Applications for service are noncancellable unless the Company otherwise agrees. Where the Company permits Customer to cancel an application for service prior to the start of service or prior to any special construction, no charges will be imposed except for those specified below.
- 2.5.7.2 Where, prior to cancellation by the Customer, the Company incurs any expenses in installing the service or in preparing to install the service that it otherwise would not have incurred, a charge equal to the costs the Company incurred, less net salvage, shall apply, but in no case shall this charge exceed the sum of the charge for the minimum period of services ordered, including installation charges, and all charges others levy against the Company that would have been chargeable to the Customer had service begun (all discounted to present value at six percent).
- 2.5.7.3 Where the Company incurs any expense in connection with special construction, or where special arrangements of facilities or equipment have begun, before the Company receives a cancellation notice, a charge equal to the costs incurred, less net salvage, applies. In such case, the charge will be based on such elements as the cost of the equipment, facilities, and material, the cost of installation, engineering, labor, and supervision, general and administrative expense, other disbursements, depreciation, maintenance, taxes, provision for return on investment, and any other costs associated with the special construction or arrangements.
- 2.5.7.4 The special charges described in 2.5.7.2 and 2.5.7.3 will be calculated and applied on a case-by-case-basis.

2.5.8 Changes in Service Requested

2.5.8.1 If the Customer makes or requests material changes in circuit engineering, equipment specifications, service parameters, premises locations, or otherwise materially modifies any provision of the application for service, the Customer's installation fee shall be adjusted accordingly.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

2. <u>REGULATIONS</u> (Cont'd)

2.6 <u>Allowances for Interruptions in Service</u>

Interruptions in service, which are not due to the negligence of, or noncompliance with the previsions of this Price List by, the Customer or the operation or malfunction of the facilities, power or equipment provided by the Customer, will be credited to the Customer as set forth in 2.6.1 for the part of the service that the interruption affects.

2.6.1 Credit for Interruptions

- 2.6.1.1 A credit allowance will be made when an interruption occurs because of a failure of any component furnished by the Company under this Price List. An interruption period begins when a Customer reports a service, facility or circuit to be interrupted and release it for testing and repair. An interruption period ends when the service, facility or circuit is inoperative but declines to release it for testing and repair, it is considered to be impaired, but not interrupted.
- 2.6.1.2 For calculating credit allowances, every month is considered to have 30 days. A credit allowance is applied on a pro rata basis against the rates specified thereunder and is dependent upon the length of the interruption. Only those facilities on the interrupted portion of the circuit will receive a credit.
- 2.6.1.3 A credit allowance will be given for interruptions of 15 minutes or more.

 Credit allowances shall be calculated as follows:

Interruptions of 24 hours or Less:

Length of Interruption	Credit Period to be Credited
Less than 15 minutes 15 minutes up to but not including 3 hours 3 hours up to but not including 6 hours 6 hours up to but not including 9 hours 9 hours up to but not including 12 hours 12 hours up to but not including 15 hours 15 hours up to but not including 24 hours	None 1/10 Day 1/5 Day 2/5 Day 3/5 Day 4/5 Day One Day

Two or more interruptions of 15 minutes or more during any 24-hour period shall be considered as one interruption.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005

2. <u>REGULATIONS</u> (Cont'd)

2.6 <u>Allowances for Interruptions in Service</u>

2.6.1 <u>Credit for Interruptions (Cont'd)</u>

2.6.1.4 For Switched Access Service, no credit will be allowed for an interruption of less than 24 hours. After the first 24 hour period, a credit equal to 1/30 of the applicable recurring transport charges will be applied to each interruption which is in excess of 12 hours and up to 24 hours.

<u>Interruptions Over 24 Hours and Less Than 72 Hours.</u>

Interruptions over 24 hours and less than 72 hours will be credited 1/5 day for each 3-hour period or fraction thereof. No more than one full days credit will be allowed for any period of 24 hours.

Interruptions Over 72 Hours.

Interruptions over 72 hours will be credited 2 days for each full 24-hour period. No more than 30 days credited will be allowed for any one month period.

2.6.2 <u>Limitations on Allowances</u>

No credit allowance will be made for:

- (a) interruptions due to the negligence of, or noncompliance with the provisions of this Price List by, the Customer, authorized user, joint user, or other common company providing service connected to the service of the Company;
- (b) interruptions due to the negligence of any person other than the Company, including but not limited to the Customer or other common companies connected to the Company's facilities;
- (c) interruptions due to the failure or malfunction of non-company equipment;
- (d) interruptions of service during any period in which the Company is not given full and free access to its facilities and equipment for the purpose of investigating and correcting interruptions;
- (e) interruptions of service during a period in which the Customer continues to use the service on an impaired basis;

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005

2. REGULATIONS (Cont'd)

2.6.2 Limitations on Allowances (Cont'd)

- (f) interruptions of service during any period when the Customer has released service to the Company for maintenance purposes or for implementation of a Customer order for a change in service arrangements, and
- (g) interruption of service due to circumstances or causes beyond the control of Company.
- 2.6.2.I Use of alternative service provided by the Company: Should the Customer elect to use an alternative service provided by the Company during the period that a service is interrupted, the Customer must pay the Price Listed rates and charges for the alternative service used.

(D) | | | | | | |

2.7 Cancellation of Service

If a customer cancels a Service Order or terminates services before the completion of the term for any reason whatsoever. Customer agrees to pay to Company the following sums which shall become due and owing as of the effective date of the cancellation or termination and be payable within 1) all Non-Recurring charges reasonable expend by Company to establish service to incurred and paid to third parties by Company on behalf of Customer, plus 3) all Recurring Charges specified in the applicable Service Order Price List for the balance of the then current term.

The terms and conditions specified in Section 3.2.3 will apply for cancellation of an access service request.

2.8 Transfers and Assignments

Kelly Faul - Regulatory Affairs Director

13865 Sunrise Valley Dr. Herndon, VA 20171

Neither the Company nor the customer may assign or transfer its rights or duties in connection with the services and facilities provided by the Company without the written consent of the other party, except that the company may assign its rights and duties (a) to any subsidiary, parent company or affiliate of the Company, (b) pursuant to any sale or transfer of substantially all the assets of the Company; or (c) pursuant to any financing, merger or reorganization of the Company.

(C)

Issued: June 7, 2013

Effective: June 30, 2013

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

2. <u>REGULATIONS</u> (Cont'd)

2.9 <u>Notices and Communications</u>

- 2.9.1 The Customer shall designate on the Service order an address to which the Company shall mail or deliver all notices and other communications, except that Customer may also designate a separate address to which the Company's bills for service shall be mailed.
- 2.9.2 The Company shall designate on the Service Order an address to which the Customer shall mail or deliver all notices and other communications, except that Company may designate a separate address on each bill for service to which the Customer shall mail payment on that bill.
- 2.9.3 All notices or other communications required to be given pursuant to this Price List will be in writing. Notices and other communications of either party, and all bills mailed by the Company, shall be presumed to have been delivered to the other party on the third business day following placement of the notice, communication or bill with the U.S. Mail or a private delivery service, prepaid and properly addressed, or when actually received or refused by the addressee, whichever occurs first.
- 2.9.4 The Company or the Customer shall advise the other party of any changes to the addresses designated for notices, other communications or billing, by following the procedures for giving notice set forth herein.

3. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE*

- 3.1 General: This section sets forth the regulations and order related charges for Access Service Requests (ASR) for Switched and Special Access Service*, as defined in this Price List. These charges are in addition to other applicable charges set forth in other sections of this tariff. In those cases where the Customer has used Company's Switched Access Service without the submission of an ASR, the Customer is responsible for the payment for such services under the terms and conditions of this tariff.
 - 3.1.1 Ordering Conditions: All services offered under this Price List will be ordered using ASR. The format of the ASR will be as specified in the industry Access Service Order Guidelines, unless otherwise specified herein. A Customer may order any number of services of the same type and between the same Premises on a single ASR. All details for services for a particular order must be identical.

The Customer shall provide all information necessary for the Company to provide and bill for the requesting service. When placing an order for Access Service, the Customer shall provide the following minimum information:

- a. Customer name and Premise(s) address(s);
- b. Billing name and address (when different from Customer name and address);
- Customer contact name(s) and telephone number(s) for the following provisioning activities: order negotiating, order confirmation, interactive design, installation and billing.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

Issued: April 21, 2020 Effective: May 8, 2020

Kelly Faul – Senior Manager 22001 Loudoun County Pkwy Ashburn, VA 20147

(N)

(N)

(C)

(C)

3. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE*

(C)

3.1 <u>General</u> (Cont'd)

3.1.1 Ordering Conditions (Cont'd)

The order date (Application Date) is the date on which the Company receives a firm commitment and sufficient information from the Customer to allow processing of the ASR. The Customer is advised of the critical events in the provisioning process, the Application Date, the Plant Test Date and the Service Commencement Date, at the time the company gives the Customer a Firm Order Confirmation (FOC). The FOC is forwarded to the Customer within 2 business days after the date on which all information needed to process the ASR has been received by the Company.

3.1.2 <u>Provision of Other Services</u> Unless otherwise specified herein, all services offered under this Price List shall be ordered with an ASR. In those cases where the Customer has used Company's Services or has used additional services without the submission of an ASR, the Customer is responsible for payment for such services under the terms and conditions of this tariff.

With the agreement of the Company, other services may subsequently be added to the ASR at any time, up to an including service date for the Access Service. When added subsequently, charges for a Design Change as set forth in Section 6.3.3.2 will apply when an engineering review is required.

Additional Engineering is not an ordering option, but will be applied to an ASR when the Company determines that Additional Engineering is necessary to accommodate a Customer request. Additional Engineering will be provided by the Company at the request of the Customer only when a Customer requests additional technical information after the Company has already provided the technical information included on the Design Layout Report as set forth herein. The Customer will be notified when Additional Engineering is required, and will be furnished with a written statement setting forth the justification for the Additional Engineering as well as an estimate of the charges. If the Customer agrees to the Additional Engineering, a firm order will be established. If the Customer does not want the service or facilities after being notified by the Company that Additional Engineering is required, the Customer may cancel the order and no charges will apply.

3.2 Access Order: An ASR may be required by the Company to provide a Customer both Switched and Special Access Service*, as described herein. An ASR will be required for each new similar service arrangement or group of common circuits. In those cases where the Customer has used Company's Switched and Special Access Service* or has obtained new similar services without the submission of an ASR, the Customer is responsible for the payment for such services under the terms and conditions of this tariff.

When a customer requests new or additional Switched Access Service, one or more ASR's may be required. The number of orders required is dependent on the type of services and/or facilities being requested.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

(N) (N)

(C)

(C)

Issued: April 21, 2020 Effective: May 8, 2020

3. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE*

(C)

3.2 <u>Access Order (Cont'd)</u>

When placing an order for either Direct Connect Service or Tandem Connect Service, as described in Sections 5.2.3.1.1 and 5.2.3.1.2, respectively, the Customer shall provide all standard ASR ordering information as specified in industry guidelines. The customer will also be required to provide this information to order additional service for an existing service type. For new Customers ordering Tandem Connect Service, the Customer will only be required to complete an ASR for installation of new service.

- 3.2.1 <u>Access Service Date Intervals</u>: Access Service is provided with one of the following Service Date intervals:
 - Standard Interval
 - Negotiated Interval

The Company will specify a FOC and the Service Commencement Date contingent on the ASR being complete as received. To the extent the Access Service can be made available with reasonable effort, the Company will provide the Access Service in accordance with the Customer's requested interval, subject to the following conditions:

3.2.1.1 <u>Standard Interval</u>: The Standard Interval for Switched and Special Access Service* will be 10 business days from the Application Date. This interval only applies to standard service offerings for a Customer which is On-Net and at locations where there are pre-existing facilities to the Customer premises. Access Services provided under the Standard Interval will be installed during Company business hours.

(C)

- 3.2.1.2 <u>Negotiated Interval</u>: The Company will negotiate a Service Date Interval with the Customer when:
 - The Customer requests a Service Date before or beyond the applicable Standard Interval Service Date; or
 - There is no existing facility connecting the Customer Premises with the Company; or
 - 3) The Customer requests a service that is not considered by the Company to be a standard service offering (for example, if Additional Engineering is required to complete the order); or
 - The Company determines that Access Service cannot be installed within the Standard Interval.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

(N) (N)

Issued: April 21, 2020 Effective: May 8, 2020

3. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE*

(C)

(C)

- 3.2 <u>Access Order: (Cont'd)</u>
 - 3.2.1 <u>Access Service Date Intervals</u>: (Cont'd)

The Company will offer a Service Date based on the type and quantity of Access Services the Customer has requested. The Negotiated Interval may not exceed by more than six months the Standard Interval Service Date, or, when there is no Standard Interval, the Company offered Service Date. All services for which rates are applied on an Individual Case Basis are provided with a Negotiated Interval.

3.2.2 Access Service Request Modifications: The Customer may request a modification of its ASR or Service prior to the Service Commencement Date. All modifications must be in writing using the industry ASR process. The Company, in its sole discretion, may accept a verbal modification from the Customer. The Company will make every effort to accommodate a requested modification when it is able to do so with the normal work force assigned to complete such an order within normal business hours. Charges for access service order modification will apply as set forth below, on a per occurrence basis.

Any increase in the number of Special Access Service* Channels, Switched Access Service lines, trunks, transport facilities, Out of Band Signaling connections or any change in engineering or functionality of a service will be treated as a new ASR with a new Service Date Interval.

3.2.2.1 Service Commencement Date Changes: ASR service dates for the installation of new services or rearrangement of existing services may be changed, but the new service date may not exceed the original Service Commencement Date by more than 30 calendar days. When, for any reason, the Customer indicates that service cannot be accepted for a period not to exceed 30 calendar days, and the Company accordingly delays the start of service, a Service Change Charge will apply. In addition, when the Customer submits a request for a Service Date Change that is less than five business days from the date of notification by the Customer, a Service Date Charge and an Expedite Charge will apply. No Expedite Charge will apply if the Customer requests a Service Date Change that is more than 5 business days from the date of request by the Customer but earlier than the original requested Service Commencement Date.

If the Customer requested service date is more than 30 calendar days after the original service date, the order will be canceled by the Company on the 31st day. Appropriate cancellation charges will be applied. If the Customer still requires the service, the Customer may be required to place a new ASR with the Company.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

(N) (N)

Issued: April 21, 2020 Effective: May 8, 2020

3. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE*

(C)

- 3.2 <u>Access Order: (Cont'd)</u>
 - 3.2.2 <u>Access Service Request Modifications: (Cont'd)</u>

The Service Date Change Charge will apply on a per order, per occurrence basis for each service date changed. The applicable charges are set forth in Section 6.3.6.2.

3.2.2.2 <u>Design Change Charge</u>: The Customer may request a Design Change to the service ordered. A Design Change is any change to an ASR which requires Engineering Review. An Engineering Review is a review by Company personnel of the service ordered and the requested changes to determine what change(s) in the design, if any, are necessary to meet the Customer's request. Design Changes include such changes as the addition or deletion of optional features or functions, a change in the type of Transport Termination (Switched Access only) or type of Channel interface. Any other changes are not considered Design Changes for purpose of this subsection and will require issuance of a new ASR and the cancellation of the original ASR with appropriate cancellation charges applied.

The Design Change Charge will apply on a per order, per occurrence basis, for each order requiring a Design Change. The applicable charges, as set forth in Section 6.3.3.2, are in addition to any Service Date Change Charges that may apply.

3.2.2.3 Expedited Order Charge: When placing an Access Order for service(s) for which a Standard Interval exists, a Customer may request a Service Commencement date that is earlier than the Standard Interval Service Date, in which case an Expedite Charge will apply. The Expedite Charge will not apply if the new Service Commencement Date is more than five days from the date of the request to the Company of the expedited order request. The request for an earlier service date may be received from the Customer prior to the service date. The Company has the exclusive right to accept or deny the Expedite Order request. However if, upon reviewing availability of equipment and scheduled work load, the Company agrees to provide service on an expedited basis and the Customer accepts the Company's proposal, an Expedite Charge will apply.

If the Company is subsequently unable to meet an agreed upon expedited service date, then the Expedite Charge will not apply.

In the event the Company provides service on a expedited basis on the Customer's request, and the Customer delays service or is not ready for delivery of service at the time of installation, a Service Date Change Charge will apply in addition to the Expedite Charge.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

(N) (N)

•

Issued: April 21, 2020 Effective: May 8, 2020

3. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE*

3.2 <u>Access Order: (Cont'd)</u>

3.2.2 <u>Access Service Request Modifications: (Cont'd)</u>

In the event that the Customer cancels an expedite request, the Expedite Charge will be added to any applicable Cancellation Charge specified herein.

In the event that the Customer requests a Service Date Change after the Company has received the original expedite request, the Expedite Charge will still apply.

An Expedite Charge will not be applied to orders expedited for Company reasons.

If costs other than additional administrative expenses are to be incurred when the Access Order is expedited, the regulations and charges for Special Construction as set forth in this Price List will apply.

The Expedited Order Charge will apply on a per order, per occurrence basis, as specified in Section 6.3.2.

3.2.3 <u>Cancellation of an Access Service Request:</u> A Customer may cancel an ASR for the installation of Switching Access Service at any time prior to notification by the Company that service is available for the Customer's use. If the Service is being provided without an ASR, a Customer may cancel the Service at any time prior to use of the Service. The cancellation date is the date the Company receives written or verbal notice from the Customer that the order or service is to be canceled. The verbal notice must be followed by written confirmation within 10 days. A Customer may negotiate an extension of s service date of an ASR for installation of new services or rearrangement of existing service, in which case a Service Date Change Charge will apply. However, the new service date cannot exceed the originally established service date by more than 30 calendar days. On the 31st day beyond the original service date, the ASR will be canceled and the appropriate Cancellation Charge will be applied.

Except as stated herein, Cancellation Charges will apply as specified in Section 6.3.6.3.

If the cancellation occurs prior to the Company's receiving the ASR or prior to use of the service, no charges shall apply.

A Customer may cancel an ASR for the installation of Special Access Service* without incurring a charge at any time prior to the acceptance of a Negotiated Interval Service Date by the Customer. Cancellation Charges will apply for Special Access Service* if the Customer cancels more than 48 hours after the Application Date. Cancellation Charges for Expedited Orders will be applied for any order canceled from the Application Date forward.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

Issued: April 21, 2020 Effective: May 8, 2020

Kelly Faul – Senior Manager 22001 Loudoun County Pkwy Ashburn, VA 20147 (C)

(C)

(C)

(N)

3.2

(C)

(C)

INTRASTATE ACCESS SERVICES PRICE LIST

3. ORDERING OPTIONS FOR SWITCHED AND SPECIAL ACCESS SERVICE*

Access Order: (Cont'd)

3.2.3 <u>Cancellation of an Access Service Request:</u>(Cont'd)

If the Company misses a service date for a Standard or Negotiated Interval Access Order by more than 30 days due to circumstances such as acts of God, governmental requirements, work stoppages and civil disturbance, the Company shall not be liable for such delay and the Customer may cancel the ASR without incurring cancellation charges.

- 3.2.4 <u>Minimum Period of Service</u>: The minimum period for which Access Service is provided and for which charges are applicable is one month.
 - 3.2.4.1 The following changes will be treated as a discontinuance of the existing service and a request for installation of a new service. All associated Non-Recurring Charges will apply for the new service, and a new minimum period will be established:
 - (1) A change in the identity of the Customer of record;
 - (2) A move by the Customer to a different building;
 - (3) A change in type of service;
 - (4) A change in Switched Access Service Interface (i.e., DS1 or DS3);
 - (5) A change in Switched Access Service Traffic Type;
 - (6) A change in type of Special Access Service* Channel Termination;
 - (7) A change from 2-point to multipoint Special Access Service*.
 - 3.2.4.2 When Access Service is disconnected prior to the expiration of the minimum period, charges are applicable for the balance of the minimum period. The Minimum Period Charge for monthly billed services will be determined as follows:

For Switched Access Service, the charge for a month or fraction thereof is the applicable minimum monthly charge for the capacity made available to the Customer.

For Special Access Service* facilities, the charge for a month or fraction thereof is the applicable monthly charge for the service as set forth in this Price List.

All applicable Non-Recurring Charges for the service will be billed in addition to the Minimum Period Charge.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

Issued: April 21, 2020 Effective: May 8, 2020

4. SPECIAL ACCESS

4.1 General

Network Services consist of any of the services offered thereunder, either individually or in combination. Each service is offered independently of all others.

4.2 <u>Transmission Service</u>

- 4.2.1 Transmission Service is offered via the Company's facilities for the transmission of one-way and two-way communications.
- 4.2.2 Digital channels over the Company's Network are furnished for full-duplex transmission of digital signals at operating speeds as follows:

(DC A)

64 Kbps	(DS-0)
56 Kbps	(DS-OD)
19.2 Kbps	
9.6 Kbps	
4.8 Kbps	
2.4 Kbps	
1.544 Mbps	(DS-1)
44.736 Mbps	(DS-3)

Digital channels operating at speeds other than those listed above may be provided at the Company's option on an Individual Case Basis (ICB). The rates for the operating speeds outlined above are described in Section 4.2.

4.2.3 Digital channels furnished by the Company at 1.544 Mbps, interconnections to such channels and equipment's interfacing to such channels shall meet the following characteristics:

Line Rate:

1.544 Mbps + 130 ppm

Line Code 1:

Bipolar Alternate Mark Inversion (AMI)

Line Code 2:

Bipolar 8 zero substitution (B8ZS)

Line Impedance:

100 ohms +/- 5% balanced

Jitter:

The multiplexer will add not more than 0.3 time slot of rms Jitter to a DS-1 signal when looped at the DS-3 point.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005

4. SPECIAL ACCESS

4.2 <u>Transmission Service</u>

4.2.4 Digital channels furnished by the Company at 44.736 Mbps, interconnection to such channels shall meet the following technical characteristics:

Line Rate:

44.736 Mbps + 20 ppm

Line Code:

Bipolar with three zero substitution (B3ZS).

Line Impedance:

75 ohms +/- 5% unbalanced

5. <u>SWITCHED ACCESS SERVICE</u>

5.1 General

Switched Access Service, which is available to Customers for their use in furnishing their services to End Users or other customers, including carriers; provides a two-point communications path between a Customer's Premises and an End User's Premises or other customer, including another carrier's, Premises. It provides for the use of common terminating, switching and transport facilities. Switched Access Service provides the ability to originate calls from an End User's Premises location to a Customer's Premises, and to terminate calls from a Customer designated Premises to an End User's Premises.

(T)

Rates and charges are set forth in Section 6.3. The application of rates for Switched Access Service is described in Section 6.3.4.

5.2 <u>Provision and Description of Switched Access Service Agreements</u>

Switched Access Service is provided in the following service type:

5.2.1 Feature Group D (FGD) Access

FGD Access, which is available to all customers, is provisioned at the DS1 level and provides trunk-side access to Company Local Switching Center switches, with an associated uniform 10XXX Access Code for the Customer's use in originating and terminating communications. Basic FGD service will be provided with Multi-Frequency In Band Signaling (SS7 is also available as a Common Switching Option for Feature Group D). In addition, Conventional Signaling for Direct Carrier Trunk Groups is available at the Customer's option. End Users of the Customer's service may also originate calls to certain FGD Access Customers without dialing the 10XXX Access Code if the End User is presubscribed, as described herein.

The Access Code for FGD switching is a uniform Access Code of the form 10XXX. A single Access Code will be the assigned number of all FGD access provided to the Customer by the Company. No Access Code is required for calls to a customer over FGD Switched Access Service if the End User's telephone exchange service is arranged for Presubscription to that Customer, as set forth herein.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

NOV 2 - 2008

Boise, Idaho Effective: November 2, 2008

Issued: October 23, 2008 Effe

SWITCHED ACCESS SERVICE (Cont'd)

5.2 <u>Provision and Description of Switched Access Service Agreements (Cont'd)</u>

5.2.1 <u>Feature Group D (FGD) Access</u>

Where no Access Code is required, the number dialed by the Customer's End User shall be a seven or ten digit number for calls in the North American Numbering Plan (NANP), except for 00- dialed calls which are routed to the predesignated Customer. For international calls outside the NANP, a seven to twelve digit number may be dialed. The form of the numbers dialed by the Customer's End User is NXX-XXXX, 0 or 1 + NXX-XXXX, NPA + NXX-XXXX, 0 or 1 + NPA = NXX-XXXX, and when the local Switching Center is equipped for International Direct Distance Dialing (IDDD), 0 + CC + NN or 011 + CC + NN.

When the 10XXX Access Code is used, FGD switching also provides for dialing the digit 0 for access to the Customer's operator, 911 for access to the Company's emergency service, or the end-of-dialing digit (#) for cut-through access to the Customer's Premises.

In addition, End Users may originate calls by dialing the 950-XXXX Access Code specific to a particular Interchange Carrier, provided that the Interchange Carrier has subscribed to the Company's Feature Group D with 950 Access Common Switching Optional Feature. If the End User is presubscribed to that Interexchange Carrier, no Access Code is necessary.

5.2.2 Manner of Provision

Trunks used for Switched Access Service may be configured for one-way (either originating only or terminating only) or for two-way directionality. It is the Customer's responsibility to order a sufficient number of trunks of each type in order to meet its desired grade of service objective. At the Customer's request, the Company will assist the Customer in sizing Switched Access Trunk groups.

5.2.3 <u>Rate Categories</u>

The following rate categories apply to Switched Access Service:

- A. Direct Connect
- B. Tandem Connect
- C. 800 Data Base Access Service
- D. Optional Features
- 5.2.3.1 Except as stated as follows, Tandem Connect Service is provided in conjunction with the tandem provider serving the area. Charges are computed in accordance with Section 2.5.2.8 preceding (Ordering, Rating, and Billing of Access Services Where More Than One Exchange Telephone Company is Involved).
 - 5.2.3.1.1 <u>Direct Connect</u>: The Company will provide Direct Connect Service, between the Customer's Premises and the Company's Local Switching Center switch(es). This transmission path is dedicated to the use of a single Customer. DS1 and DS3 facilities are available for Direct Connect Service.

Issued: June 17, 2021 Effective: July 1, 2021

Kelly Faul – Senior Manager (T) 22000 AOL Way (T)

Dulles, VA 20166 (T)

(T)

(T)

(C)

(C)

(C)

(C)

(C)

(M)(C)

(M)(C) (M)

(N)

INTRASTATE ACCESS SERVICES PRICE LIST

5. <u>SWITCHED ACCESS SERVICE (Cont'd)</u>

5.2 <u>Provision and Description of Switched Access Service Agreements (Cont'd)</u>

5.2.3 Rate Categories (Cont'd)

A DS1 facility is capable of transmit at a nominal speed of 1.544 Mbps, with the capability to channelize up to 24 voice frequency transmission paths. A DS3 facility is capable of transmitting at a nominal speed of 44.736 Mbps, with the capability to channelize up to 672 voice-frequency transmission paths. For DS3 facilities, if the Company is required to install additional fiber optic equipment for the benefit of the Customer, then the Customer has the option to choose either an optical or electrical interface. This Direct Connect rate category is comprised of a monthly Entrance Facilities charge and a per minute of use End Office switching charge as specified in 6.3.3.4.

The Company will provide Direct Connect between the Customer's Premises and the ILEC's local switch when the local switch and the facilities are leased by the Company from the ILEC. The Company will bill the Direct Connect rates when the ILEC's Category 11 Daily Usage Feed Records indicate that the call was routed directly from the Customers' Premises to the ILEC switch serving the end user customer without routing through the access tandem.

When a Customer purchases Direct Connect, if the number of calls over the direct trunks has reached its maximum level, calls not able to be switched over the direct trunks will overflow/be routed to the Customer via an access tandem. In that event, the Customer will be assessed the Tandem Connect rates for such calls. The Direct Connect rate is comprised of the Minute of Use (MOU) based charges found in Section 6.3.3.4.2.

The Direct Connect rate is a combination of the former End Office Switching and End Office Trunk Port rate element charges.

| (M)

Issued: June 17, 2021 Effective: July 1, 2021

Some materials on this page was moved to Page 47.1.

Kelly Faul, Senior Manager (T) 22000 AOL Way (T) Dulles, VA 20166 (T)

- 5. SWITCHED ACCESS SERVICE (Cont'd)
 - 5.2 Provision and Description of Switched Access Service Agreements (Cont'd)
 - 5.2.3 Rate Categories (Cont'd)
 - 5.2.3.1.2 Tandem Connect: Tandem Connect consists of circuits from the point of interconnection with Customer's tandem provider to the Company's Local Switching Center. This Tandem Connect rate category is comprised of a Minutes of Use (MOU) based End-Office switching and tandem switched transport functions.

Tandem Connect Charges apply on a per-minute-of-use basis when calls are switched by an ILEC's tandem switch to or from the Company's Local Switching Center or are switched through a tandem switch for which the Company leases from the ILEC for the tandem switching function. The rate will vary based on whether the Company leases the tandem switching function on a call. Tandem Connect rates are found in Section 6.3.3.4.3.

Rates and charges for Tandem Connect minutes for which the Company does not lease the tandem switching function from the ILEC do not contain the ILEC's tandem switching charge and are denoted in the Switched Access Rates section as Tandem Connect Without Tandem Switching. Rates for Tandem Connect minutes for which the Company does lease the tandem switching function from the ILEC contain the ILEC's Tandem Switching Charge and are denoted in the Switched Access Rates section as Tandem Connect with Tandem Switching.

The Tandem Connect Without Tandem Switching rate is a combination of the former End Office Switching, End Office Trunk Port, Tandem Switched Transport, and Tandem Switched Transport Facility (10 miles) rate element charges. The Tandem Connect With Tandem Switching rate is a combination of the former End Office Switching, End Office Trunk Port, Tandem Switched Transport, Tandem Switched Transport Facility (10 miles), and Tandem Switching rate element charges.

Some material on this page was moved from Page 47.

Issued: June 17, 2021

Effective: July 1, 2021

Kelly Faul, Senior Manager 22000 AOL Way Dulles, VA 20166

 (\mathbf{M}) $(\mathbf{M})(\mathbf{C})$

(M)

(C)

(C)

(N)

(M)(T)

(M)(T)

(M)

(M)(T)(M)

(M)

(M)(T)

(M)

(M)

(C) (T)

(N)

INTRASTATE ACCESS SERVICES PRICE LIST

5. SWITCHED ACCESS SERVICE (Cont'd)

5.2 Provision and Description of Switched Access Service Agreements (Cont'd)

5.2.3 Rate Categories (Cont'd)

5.2.3.1.3 800 Data Base Access Service

800 Data Base Access Service is a service offering utilizing originating Trunk side Switched Access Service. When an Toll Free SAC*+NXX+XXXX call is originated by an End User, the Company will perform Customer Identification based on screening of the full ten-digits of the Toll Free number to determine the Customer location to which the call is routed.

The 800 Data Base charge, which consists of a single, fixed rate element, applies on a per query basis.

5.2.3.1.4 Switched Access Service Optional Features

Nonchargeable Optional Features: 5.2.3.1.4.1

Where transmission facilities permit, the Company will, at the option of the Customer, provide the following nonchargeable optional feature, in association with Switched Access Service.

Supervisory Signaling (a)

5.2.3.1.4.2 Chargeable Optional Features: Where transmission facilities permit, the Company will, at the option of the Customer, provide the following chargeable optional features, as described in Section 5.5.2, in association with Switched Access Service.

(a) Signaling Transfer Point Access

5.2.3.1.4.3 Feature Group D Optional Features

> Following are the various optional features that are available in lieu of, or in addition to, the standard features provided with Feature Group D. Optional features are provided as Common Switching Optional Features as described in Section 5.5.3.1.

> 5.2.3.1.4.3.1 Common Switching Optional Features: At the Customer's option, the following standard features are available at the rates specified in Section 7.4.7.1:

- (a) Alternate Traffic Routing
- (b) Automatic Number Identification (ANI)
- (c) Cut-Through
- (d) Service Class Routing
- (e) Feature Group D with 950 Access
- (f) Signaling System Seven (SS7)
- (g) Basic Initial Address Message Delivery

Some material on this page was moved from Page 47.

Effective: July 1, 2021

Issued: June 17, 2021

Kelly Faul – Senior Manager (T) 22000 AOL Way (T) Dulles, VA 20166

Idaho Public Utilities Commission Office of the Secretary ACCEPTED FOR FILING July 1, 2021 Boise, Idaho

5. <u>SWITCHED ACCESS SERVICE (Cont'd)</u>

- 5.2 Provision and Description of Switched Access Service Agreements (Cont'd)
 - 5.2.3 Rate Categories (Cont'd)
- (h) Called Directory Number Delivery
- (I) Flexible Automatic Number Identification Delivery

5.2.4 Billing Validation Service:

The Company shall arrange to have its billing validation data stored in one of the existing Line Information Databases (LIDB). It will be the responsibility of the Customer to identify this database through established industry procedures and to query the billing validation data in the LIDB. Based on the received query information, the LIDB will respond with an SS7 formatted confirmation of validity or denial for the requested billing option. Access to LIDB provides Customers with potential toll fraud detection.

The LIDB will contain a record for every working line number and Billed Number Group served by the Company.

The Company will update the LIDB information on a daily basis.

LIDB service is provided on an on-line, call-by-call basis. Company data accessed from the LIDB shall remain the sole property of the Company and may not be stored or reproduced by the customer for any reason.

The Company will have procedures in place to deactivate billing validation data in the event that it is being used fraudulently.

5.2.5 <u>Design Layout Report:</u>

At the request of the Customer, the Company will provide to the Customer the makeup of the facilities and services provided from the Customer's Premises to the first point of switching. This information will be provided in the form of a Design Layout Report. The Design Layout Report will be provided to the Customer at no charge.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

5. <u>SWITCHED ACCESS SERVICE (Cont'd)</u>

5.2 Provision and Description of Switched Access Service Agreements (Cont'd)

5.2.6 Acceptance Testing:

At no additional charge, the Company will, at the Customer's request, cooperatively test, at the time of installation, the following parameters: loss, C-notched noise, C-message noise, 3-tons slope, d.c. continuity and operational signaling.

5.2.7 Ordering Options and Conditions:

Access Service is ordered under the Access Order provisions set forth in Section 3.2. Also included in that section are other charges which may be associated with ordering Switched Access Service.

5.2.8 Competitive Pricing Arrangements:

Competitive pricing arrangements for Local Transport - Entrance Facilities and Local Transport - Direct Trunked Transport can be furnished to meet the communications needs of specific Customers on a case by case basis under individual contracts. The competitive pricing arrangement contracts, once executed, will be filed with the Idaho Public Service Commission under proprietary agreement.

5.3 <u>Obligations of Company</u>

In addition to the obligations of the Company set forth in other sections of this Price List, the Company has certain other obligations concerning the provision of Switched Access Service. These obligations are as follows:

5.3.1 <u>Network Management</u>

The Company will administer its Network to insure the provision of acceptable service levels to all telecommunications users of the Company's Network Services. Generally, service levels are considered acceptable only when both End Users and Customers are able to establish connections with little or no delay encountered within the Company Network. The Company reserves the right to apply protective controls, (i.e., those actions, such as call gapping, which selectively cancel the completion of traffic), over any traffic carried over its Network, including that associated with a Customer's Switched Access Service. Generally, such protective measures would only be taken as a result of occurrences such as failure or overload of Company or Customer facilities, natural disasters, mass calling or national security demands. The Customer will notify the Company of anticipated peaked services as stated below. Based on the information provided, the Company will work cooperatively with the Customer to determine the appropriate level

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005

5. <u>SWITCHED ACCESS SERVICE (Cont'd)</u>

5.3 Obligations of Company (Cont'd)

5.3.1 Network Management (Cont'd)

of control. In the event that the protective controls applied by the Company result in the complete loss of service by the Customer, the Customer will be granted a credit allowance for service interruption as set forth in 2.6.

When a Customer uses the Company's facilities to offer services for which a substantial call volume or peaked service is expected during a short period of time, the Customer must notify the Company at least 24 hours in advance of each peak period. For events scheduled during weekends or holidays, the Company must be notified no later than 5:00 p.m. local time the prior business day. Notification should include the nature, time, duration, and frequency of the event, an estimated call volume, and the NPA NXX and line number(s) to be used. On the basis of the information provided, the Company may invoke network management controls if required to reduce the probability of excessive Network congestion. The Company will work cooperatively with the Customer to determine the appropriate level of such control. Failure to provide prescribed notification may result in Customer caused Network congestion, which could result in discontinuance of service under Section 2.5.5 and/or damages under Section 2.1.4.

5.4 Obligations of the Customer

In addition to obligations specified elsewhere in this Price List, the Customer has certain specific obligations pertaining to the use of Switched Access Service, as follows:

5.4.1 Report Requirements:

When a Customer orders Switched Access Service for bother interstate and intrastate use, the Customer is responsible for providing Jurisdictional Reports as set forth in Section 2.3.3 preceding. Charges will be apportioned in accordance with those reports. The method to be sued for determining the intrastate charges is set forth therein.

5.4.2 <u>Supervisory Signaling:</u>

The Customer's facilities at the premises of the ordering Customer shall provide the necessary On-Hook, Off-Hook answer and disconnect supervision.

5.4.3 <u>Design of Switched Access Services:</u>

It is the Customer's responsibility to assure that sufficient Access Services have been ordered to handle its traffic.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005 Effective: March 13, 2005

5. <u>SWITCHED ACCESS SERVICE (Cont'd)</u>

5.5 <u>Switched Access Optional Features:</u>

Following are descriptions of the various optional features that are available in lieu of, or in addition to, the standard features provided with the Feature Groups for Switched Access Service.

5.5.1 Nonchargeable Optional Feature

(a) <u>Supervisory Signaling:</u> where the transmission parameters permit, and where signaling conversion is required by the Customer to meet its signaling capacity, the Customer may order an optional supervisory signaling arrangement in the form of Multi-frequency (MF) Signaling for each transmission path.

5.5.2 Chargeable Option Features

(C) (C)

(a) Signaling Transfer Point Access: The Customer will be charged a per mile charge and a per port charge for access to a specialized switch which provides SS7 network access and performs SS7 messaging routing and screening. If a customer is connected to a third party SS7 service provider, an additional charge, as specified in Section 7.4.5.2 will apply.

5.5.3 Feature Group D Optional Features

5.5.3.1 <u>Common Switching Optional Features</u>

(a) Alternate Traffic Routing: This option provides the capability of directing originating traffic from a Local Switching Center to a direct access Trunk group, with additional traffic overflowing to the access tandem Trunk group and then to a Customer designated Premises. Multiple Customer Premises Alternate Routing is also available where originating traffic from a Local Switching Center is directed via a Trunk group to a Customer designated Premises until that group is fully loaded, and than additional originating traffic from the same Local Switching Center or access tandem is delivered via a different Trunk group to a second Customer designated Premise. The Customer shall specify the last Trunk CCS desired for the high use group.

Issued: June 17, 2021 Effective: July 1, 2021

Kelly Faul – Senior Manager (T) 22000 AOL Way (T) Dulles, VA 20166 (T)

5. <u>SWITCHED ACCESS SERVICE (Cont'd)</u>

5.5 Switched Access Optional Features:

5.5.3 <u>Feature Group D Optional Features(Cont'd)</u>

5.5.3.1 <u>Common Switching Optional Features (Cont'd)</u>

(b) Automatic Number Identification (ANI): This option provides the automatic in-band transmission signaling of a seven or ten digit number and information digits to the Customer's Premises for calls originating in the LATA for the identification of the calling station. The ANI feature is a Local Switching Center software function which is associated on a call-by-call basis with: 1) all individual transmission paths in a Trunk group routed directly between a Local Switching Center and a Customer's Premises; or where technically feasible, 2) all individual transmission paths in a Trunk group between a Local Switching Center and an Access Tandem, and a trunk group between an Access Tandem and a Customer's Premises.

The ten digit ANI telephone number is only available with Feature Group D. The ten digit ANI telephone number consists of the Numbering Plan Area (NPA) plus the seven digit ANI telephone number. The ten digit ANI telephone number will be transmitted on all calls except those identified as multi-party line or ANI failure, in which case only the NPA will be transmitted.

- (c) <u>Cut Through:</u> This option allows End Users of the Customer to reach the Customer's Premises by using the end of dialing digit (#) at the end of the dialing sequence. The Company will not record any other dialed digits on these calls.
- (d) Service Class Routing: This option provides the capability of directing originating traffic from a Local Switching Center to a Trunk group to a Customer designated Premises, based on the line class of service and service prefix indicator. A domestic Interexchange Carrier may not order more than four different routes per Local Switching Center or Access Tandem. An international Interexchange Carrier may order up to four additional routes.
- (e) Feature Group D with 950 Access: This option provides for the routing of originating calls, dialed using a 950-10XX or 950-1XXX Access Code, to the FGD Customer using FGD signaling protocols and technical specifications. The Customer is responsible for distinguishing between standard FGD calls and 950-dialed calls delivered over the same trunks.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

5. SWITCHED ACCESS SERVICE (Cont'd)

5.5 <u>Switched Access Optional Features:</u>

5.5.3 <u>Feature Group D Optional Features</u> (Cont'd)

5.5.3.1 <u>Common Switching Optional Features</u> (Cont'd)

- (f) <u>Signaling System Seven (SS7):</u> This option provides out of band transmission of SS7 protocol signaling information between the Local Switching Center switching system and the Customer's designated Premises. Prior to installation of any SS7 circuits, the Customer must agree to participate in SS7 certification testing. The Company will provide a testing plan to the Customer, and reserves the right to deny SS7 connectivity if the Customer's circuits do not meet the testing requirements.
- (g) <u>Basic Initial Address Message Delivery:</u> This option permits the following optional SS7 signaling call setup parameters: User Service Information, Called Party Number, Charge Number, Originating Line Information, Transit Network Selection, Carrier Selection, Service Code and Access Transport.
- (h) <u>Called Directory Number Delivery:</u> This option provides the customer with the telephone number to which the call was directed. The seven or ten digit number is provided as part of the in-band transmission and MF signaling. The Called Directory Number Delivery feature is associated on a call-by-call basis with all individual transmission paths in a Trunk group routed from an Access Tandem or the originating Local Switching Center. This option is available except when FGD is provided with 950 access or CutThrough features.
- (I) Flexible Automatic Number Identification Delivery: This feature is a network enhancement to ANI. The feature is available on inbound signaling or in the Originating Line Information Parameter in the Basic Initial Address Message Delivery optional feature for SS7 signaling. Flexible ANI will provide additional values for Information Indicator (ii) digits that are associated with various classes of service not associated with the standard ANI digits. This feature may only be used in conjunction with ANI. The following Information Indicator codes are available: Confinement / Detention Facility; Outward Wide Area Telecommunications Service; Cellular Service; Private Pay Station; and, Access for Private Virtual Networks.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005 Effective: March 13, 2005

SWITCHED ACCESS SERVICE (Cont'd)

5.6 Switched Access IP Direct Connect

- The Company will provide Switched Access IP Direct Connect Service, between the Customer's 5.6.1 equipment and the Company's Local Switching Center switch(es). Customer is responsible for purchasing an IP Interconnection service to access the Company's SWA IP Direct Connect Service. The application of rates for SWA IP Direct Connect Service is described in Section 6.8.
 - The SWA IP Direct Connect Service is provided to Customers for which Company has A. installed a gateway or gateways (the "Gateway") that will allow IP traffic to terminate to the Company's switched network.
 - B. The SWA IP Direct Connect Service includes connectivity from the Gateway to Company's private IP network, via a dedicated connection purchased from the Company.
 - The SWA IP Direct Connect Service allows voice and call set-up signaling to be carried C. from the Gateway to the Company's switched network for termination by Company's local switches.
 - The SWA IP Direct Connect Service may be used to terminate traffic to all valid Company D. NPA-NXXs. Calls will not be completed to 950-XXXX access codes, local operator assistance (0- and 0+), Directory Assistance (411), service codes 611, 911, and 101XXXX access codes or NPA NXXs not assigned to Company.
 - E. There are two types of rates and charges that apply to SWA IP Direct Connect Service. These are Usage Rates and Non-Recurring Charges.

Usage Rates: Usage rates are rates that are applied on a per access minute or per query basis. Usage rates are accumulated over a monthly period.

Non-Recurring Charges: Non-Recurring charges are one time charges that apply for a specific work activity (i.e., installation of new service or change to an existing service). Non-Recurring Charges apply to each SWA IP Direct Connect Service installed. The charge is applied per line or trunk.

Issued: October 5, 2012 Effective: October 15, 2012

Idaho Public Utilities Commission Office of the Secretary ACCEPTED FOR FILING

Kelly Faul - Regulatory Affairs Director

SWITCHED ACCESS SERVICE (Cont'd)

- 5.6 Switched Access IP Direct Connect (Cont'd)
 - 5.6.2 <u>Eligibility:</u> In order to be eligible to subscribe to this SWA IP Direct Connect Service, the following conditions must be met:
 - Customers must have purchased from Company and had installed a gateway or gateways, pursuant to a separate non-tariff agreement, that will allow IP traffic to terminate to the Company's switched network;
 - Customers must purchase from Company and have Company install the IP Direct Connect
 Trunk Group on the Gateway prior to commencement of SWA IP Direct Connect Service;
 - C. The Customer must have in place a private IP dedicated connection to the Gateway(s) of sufficient capacity to handle the traffic exchanged between the Customer and the Company; and
 - D. Not more than eighty-nine percent (89%) of the Customer's traffic routed to Company over the previous three (3) months via the SWA IP Direct Connect Service must be jurisdictionally intrastate.
 - 5.6.3 Expiration Date: Company shall have the right to cease offering SWA IP Direct Connect Service upon cancellation of this Section 5.6. Notwithstanding the forgoing, Company shall continue to provide service to current Customers purchasing SWA IP Direct Connect Service for three months following cancellation of Section 5.6.
 - 5.6.4 <u>Billing of SWA IP Direct Connect Service Minutes:</u> When the Customer uses IP signaling, it will transmit the telephone number of the calling party or, if different from the telephone number, the billing number of the calling party.

Issued: October 5, 2012 Effective: October 15, 2012

Kelly Faul – Regulatory Affairs Director 13865 Sunrise Valley Dr. Herndon, VA 20171

5. SWITCHED ACCESS SERVICE (Cont'd)

5.6 Switched Access IP Direct Connect (Cont'd)

5.6.5 General Regulations

- A. In order to subscribe to this offering, a Letter of Agreement for Firm Order Commitment provided by the Company must be executed with the Customer's signature and acknowledged by a Company representative's signature. The Letter of Agreement shall contain the starting date of the Customer's term and the Access Customer Name Abbreviations (ACNAs).
- B. When placing an order for SWA IP Direct Connect Service, the Customer shall provide completed and executed order forms as required by the Company.
- C. Customer agrees to purchase SWA IP Direct Connect Service for a period of no less than two years unless the Company cancels this Section 5.6 in which case Section 5.6.3 shall govern.
- D. Failure by Customer to provide required signaling information -- including, but not limited to, Calling Party Number and Automatic Number Information -- such that Company can properly determine the jurisdiction of traffic shall be grounds for Company to terminate this offering to the Customer.
- E. In order to maintain the quality of the SWA IP Direct Connect Service, the Company reserves the right to perform preventive maintenance of software updates to the network. This could result in the SWA IP Direct Connect Service being unavailable during the time period between 12:00 A.M. and 4:00 A.M. Eastern Time on any given Wednesday or Sunday. However, the Company only expects to utilize this maintenance window for any given switch on the average of once a quarter. In addition, the Company will make every reasonable effort to provide advance notice to those Customers likely to be severely affected by such maintenance work. This maintenance window may be adjusted by the Company upon written notice to the Customer.

Issued: October 5, 2012

Kelly Faul – Regulatory Affairs Director

13865 Sunrise Valley Dr. Herndon, VA 20171 Effective: October 15, 2012

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

Boise, Idaho

5. SWITCHED ACCESS SERVICE (Cont'd)

5.6 Switched Access IP Direct Connect (Cont'd)

5.6.6 Obligations of Company

In addition to the obligations of the Company set forth in other sections of this tariff, the Company has certain other obligations concerning the provision of SWA IP Direct Connect Service. These obligations are as follows:

A. Network Management

The Company will administer its Network to ensure the provision of acceptable service levels to all telecommunications users of the Company's Network Services. Generally, service levels are considered acceptable only when both End Users and Customers are able to establish connections with little or no delay encountered within the Company Network. The Company reserves the right to apply protective controls, (*i.e.*, those actions, such as call gapping, which selectively cancel the completion of traffic), over any traffic carried over its Network, including that associated with a Customer's SWA IP Direct Connect Service. Generally, such protective measures would only be taken as a result of occurrences such as failure or overload of Company or Customer facilities, natural disasters, mass calling or national security demands. The Customer will notify the Company of anticipated peaked services as stated below. Based on the information provided, the Company will work cooperatively with the Customer to determine the appropriate level of control. In the event that the protective controls applied by the Company result in the complete loss of service by the Customer, the Customer will be granted a credit allowance for service interruption as set forth in Section 2.6.

Issued: October 5, 2012

Effective: October 15, 2012
Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

SWITCHED ACCESS SERVICE (Cont'd)

5.6 Switched Access IP Direct Connect (Cont'd)

5.6.7 Obligations of the Customer

In addition to obligations specified elsewhere in this tariff, the Customer has certain specific obligations pertaining to the use of SWA IP Direct Connect Service, as follows:

Report Requirements:

When a Customer orders SWA IP Direct Connect Service for both interstate and intrastate use, the Customer is responsible for providing Jurisdictional Reports as set forth in Section 2.3.3 preceding. Charges will be apportioned in accordance with those reports. The method to be used for determining the intrastate charges is set forth therein Section 2.3.3.

B. Supervisory Signaling:

The Customer's premises equipment shall provide the necessary On-Hook, Off-Hook answer and disconnect supervision.

C. Design of SWA IP Direct Connect Services:

It is the Customer's responsibility to assure that sufficient SWA IP Direct Connect Service has been ordered to handle its traffic.

D. Network Management

Kelly Faul - Regulatory Affairs Director

13865 Sunrise Valley Dr.

When a Customer uses the Company's facilities to offer services for which a substantial call volume or peaked service is expected during a short period of time, the Customer must notify the Company at least 24 hours in advance of each peak period. For events scheduled during weekends or holidays, the Company must be notified no later than 5:00 p.m. local time the prior business day. Notification should include the nature, time, duration, and frequency of the event, an estimated call volume, and the NPA NXX and line number(s) to be used. On the basis of the information provided, the Company may invoke network management controls if required to reduce the probability of excessive Network congestion. The Company will work cooperatively with the Customer to determine the appropriate level of such control. Failure to provide prescribed notification may result in Customer caused Network congestion, which could result in discontinuance of service under Section 2.5.6.

Issued: October 5, 2012

Effective: October 15, 2012

Idaho Public Utilities Commission Office of the Secretary ACCEPTED FOR FILING

6. RATES AND CHARGES

6.1 Special Construction

6.1.1 Basis for Rates and Charges

Rates and Charges for special construction will be based on the costs incurred by the Company and may include: (1) nonrecurring type charges; (2) recurring type charges; (3) termination charges; or (4) a combination thereof.

6.1.2 Basis for Cost Computation)

The cost referred to in 6.1.1 may include one or more of the following items to the extent that they are applicable.

- (a) Cost installed of the facilities to be provided, including estimated costs for the rearrangement of existing. Cost installed includes the cost of:
 - (1) equipment and materials provided or used,
 - (2) engineering, labor and supervision,
 - (3) transportation, and
 - (4) rights-of-way;
- (b) cost of maintenance;
- depreciation on the estimated cost installed of any facilities provided, based on the anticipated useful service life of the facilities with an appropriate allowance for the estimated net salvage;
- (d) administration, taxes and uncollectable revenue on the basis of reasonable average costs for these items;
- (e) license preparation, processing and related fees;
- (f) Price List preparation, processing and related fees;
- (g) any other identifiable costs related to the facilities provided; or
- (h) an amount for return and contingencies.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

6. <u>RATES AND CHARGES (Cont'd)</u>

- 6.1 <u>Special Construction (Cont'd)</u>
 - 6.1.3 Termination Liability

To the extent that there is no other requirement for use by the Company, a termination liability may apply for facilities specially constructed at the request of the Customer.

- 6.1.3.1 The termination liability period is the estimated service life of the facilities provided.
- 6.1.3.2 The amount of the maximum termination liability is equal to the estimated amounts for:
 - (a) cost installed of the facilities provided including estimated costs for rearrangements of existing facilities and/or construction of new facilities as appropriate, less net salvage. Cost installed including the cost of:
 - (1) equipment and materials provided or used,
 - (2) engineering, labor and supervision,
 - (3) transportation, and
 - (4) rights-of-way;
 - b) license preparation, processing and related fees;
 - (c) Price List preparation, processing and related fees;
 - (d) cost of removal and restoration, where appropriate; and
 - (e) any other identifiable costs related to the specially constructed or rearranged facilities.
- 6.1.3.3 The applicable termination liability charge is based on the normal method for circulating the unpaid balance of a term obligation.

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

6. <u>RATES AND CHARGES (Cont'd)</u>

6.2 <u>Special Access Service*</u>

6.2.1 General

Rates are composed of three elements which may apply to a Customer's service, depending upon the specific service requested and its location.

- 6.2.1.1 The channel termination rate element provides for the termination of the communications path at the Customer designated location. One channel termination charge applies for each Customer designated location at which a channel is terminated.
- 6.2.1.2 The channel mileage rate element is determined by the Vertical and Horizontal Coordinates ("V & H") method, as set forth in the National Exchange Company Association Price List, F.C.C. No 4. When the calculation results in a fraction of a mile, the total is rounded up to the next whole mile before applying the rate.
- 6.2.1.3 Optional Features for which charges are applied only include multiplexing.

6.2.2 <u>Voice Grade Service</u>

There are two types of Voice Grade Services. The service is compatible with either 2-wire ground start and loop start equipment or 4-wire E/M signaling equipment. 4-wire supports analog data transmission speeds of up to 19.2 Kbps. The rates for Voice Grade channels as described in 4.2.2 are as follows:

5.2.2.1 This service consists of making Voice Grade capacity available in a 24-hour per day, 7 days per week basis.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

Issued: April 21, 2020 Effective: May 8, 2020

Kelly Faul – Senior Manager 22001 Loudoun County Pkwy Ashburn, VA 20147 (N) (N)

(C)

6. RATES AND CHARGES (Cont'd)

6.2 <u>Special Access Service* (Cont'd)</u>

(C)

6.2.2 <u>Voice Grade Service (Cont'd)</u>

6.2.2.2 <u>Voice Grade Service Rates:</u>

RECURRING PER MONTH

Channel Termination per point of termination

2 wire voice/analog data
 4 wire voice/analog data
 ICB

NON RECURRING CHARGE

Channel Termination, per point of termination

- 2 wire voice/analog data ICB subsequent, same location ICB

- 4 wire voice/analog data ICB

- subsequent, same location ICB

In addition to Channel Termination Charges, Channel Mileage Charges (both recurring and non-recurring) may apply as determined on an individual case basis.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

Effective: May 8, 2020

Kelly Faul – Senior Manager 22001 Loudoun County Pkwy Ashburn, VA 20147

Issued: April 21, 2020

6. RATES AND CHARGES (Cont'd)

6.2 <u>Special Access Service* (Cont'd)</u>

6.2.3 <u>Digital Data Service</u>

Digital Data Service is provided at transmission rates of 2.4, 4.8, 9.6, 56 and 64 Kbps.

6.2.3.1 This service consists of making DS-0 capacity available on a 24-hour per day, 7 says per week basis.

6.2.3.2 Digital Data Services

Channel Termination Per point of termination	RECURRING PER MONTH
2.4 Kbps	ICB
4.8 Kbps	ICB
9.6 Kbps	ICB
19.2 Kbps	ICB
56 Kbps	ICB
64 Kbps	ICB

In addition to Channel Termination Charges, Channel Mileage Charges (both recurring and non-recurring) may apply as determined on an individual case basis.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

(N) (N)

(C)

Kelly Faul – Senior Manager 22001 Loudoun County Pkwy Ashburn, VA 20147

Issued: April 21, 2020

Effective: May 8, 2020

6. <u>RATES AND CHARGES (Cont'd)</u>

6.2 <u>Special Access Service*</u> (Cont'd)

6.2.3 <u>Digital Data Service</u>

Channel Termination, per point of termination	NON-RECURRING
per point of termination	NON-RECORDING
2.4 Kbps	ICB
4.8 Kbps	ICB
9.6 Kbps	ICB
19.2 Kbps	ICB
56 Kbps	ICB
64 Kbps	ICB
Subsequent, same location	
2.4 Kbps	ICB
4.8 Kbps	ICB
9.6 Kbps	ICB
19.2 Kbps	ICB
56 Kbps	ICB
64 Kbps	ICB

Effective: May 8, 2020

Kelly Faul – Senior Manager 22001 Loudoun County Pkwy Ashburn, VA 20147

Issued: April 21, 2020

(N)

^{*}Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

6. RATES AND CHARGES (Cont'd)

6.2 <u>Special Access Service*</u> (Cont'd)

6.2.4 <u>DS-1 Service</u>

DS-1 service is a digital transmission facility of 1.544 Mbps with a capacity of up to 24 analog or digital channels. This service supports voice, analog data, digital data and video.

6.2.4.1 This service consists of making DS-1 capacity available 24-hours per day, 7 days per week.

6.2.4.2 <u>DS-1 Rates</u>

.

- multiplexing, DS-1 to DS-0

RECURRING PER MONTH

ICB

Channel Termination	
per point of termination	ICB
Subsequent, same location	ICB

Channel Mileage	
- first mile	ICB
- each additional mile	ICB

Optional Features - multiplexing, DS-1 to DS-0 ICB

NON RECURRING CHARGES

Channel termination per point of termination	ICB
Subsequent, same locations	ICB
Ontional Features	

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

Issued: April 21, 2020 Effective: May 8, 2020

Kelly Faul – Senior Manager 22001 Loudoun County Pkwy Ashburn, VA 20147

6. RATES AND CHARGES (Cont'd)

6.2 <u>Special Access Service*</u> (Cont'd)

6.2.5 <u>DS-3 Service</u>

DS-3 service is a digital transmission facility of 44.736 Mbps with a capacity of 28 DS-1 channels or 672 Voice, Analog Data or Digital Data Channels.

6.2.5.1 The service consists of making DS-3 capacity available 24-hours per day, 7 days per week.

6.2.5.2 DS-3 Rates

Channel Termination

per point of termination ICB Subsequent, same location ICB

Channel Mileage

- 0 miles ICB - over 0 miles ICB

Optional Features

- multiplexing, DS-3 to DS-1 ICB

NON RECURRING CHARGES

Channel Termination

per point of termination ICB Subsequent, same location ICB

Optional Features

- multiplexing, DS-3 to DS-1 ICB

Effective: May 8, 2020

Kelly Faul – Senior Manager 22001 Loudoun County Pkwy Ashburn, VA 20147

Issued: April 21, 2020

^{*}Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

6. RATES AND CHARGES (Cont'd)

6.2 <u>Special Access Service*</u> (Cont'd)

6.2.6 Service Calls

When a customer reports trouble to the Company for clearance and no trouble is found in the Company's facilities, the Customer may be responsible for payment of a charge calculated from the time Company's personnel are dispatched to the Customer Premise until the work is completed.

6.2.6.1 <u>Service Call Charge Rates.</u> <u>Per hour rate</u>

per technician \$95.00

6.2.7 <u>Individual Case Basis Arrangements</u>

The Company may, in response to a bona fide request from a potential customer or User, develop a bid for services offered in this Price List, or a bid for a Special Service Arrangement not currently offered by the Company. Prices quoted in response to such requests may be different from those in effect in this Price List. An Individual Case Basis arrangement will be offered to the subscriber for acceptance in writing. Such Individual Case Basis Arrangements will specify, among other things, the length of service, minimum volume of service required, and rate and charges for the proposed service.

*Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

Issued: April 21, 2020 Effective: May 8, 2020

Kelly Faul – Senior Manager 22001 Loudoun County Pkwy Ashburn, VA 20147

6. <u>RATES AND CHARGES (Cont'd)</u>

6.3 <u>Switched Access Services</u>

There are three types of rates and charges that apply to Switched Access Service. These are Monthly Recurring Charges, usage rates and Non-Recurring Charges.

<u>Monthly Recurring Charges</u>: Monthly Recurring Charges are flat rates for facilities that apply each month or fraction thereof that a specific rate element is provided.

<u>Usage Rates</u>: Usage rates are rates that are applied on a per access minute or per query basis. Usage rates are accumulated over a monthly period.

<u>Non-Recurring Charges</u>: Non-Recurring charges are one time charges that apply for a specific work activity (i.e., installation of new service or change to an existing service).

(a) <u>Installation of Service</u>: Non-Recurring charges apply to each Switched Access Service installed. The charge is applied per line or trunk.

6.3.1 <u>Application of Rates</u>

- 6.3.1.1 <u>Direct Connect</u> The Direct Connect rate is assessed based on the total of the monthly Entrance Facilities charge and the usage rates of applicable rate elements. The monthly facilities charge consists of a fixed rate based on the type of the facilities, i.e., DS1 or DS3; and any applicable usage rates.
- 6.3.1.2 <u>Tandem Connect</u> The Tandem Connect rate is assessed based on the usage charges for End-Office switching and Tandem Switched Transport functions.
- 6.3.1.3 <u>800 Data Base Access Service Charge</u> The 800 Data Base Access Service Charge applies for the translation of a specific ten-digit Toll Free number to a Customer Identification on a per query basis.
- 6.3.2 <u>Billing of Access Minutes:</u> When recording originating calls over FGD with multi-frequency address signaling, usage measurement begins when the first wink supervisory signal is forwarded from the Customer's facilities. The measurement of originating call usage over FGD ends when the originating FGD entry switch receives disconnect supervision from either the originating End User's Local Switching Center (indicating that the originating End User has disconnected), or the Customer's facilities, whichever is recognized first by the entry switch.

For terminating calls over FGD with multi-frequency address signaling, the measurement of access minutes begins when a seizure signal is received from the Carrier's Trunk group at the Point of Presence within the LATA. The measurement of terminating call usage over FGD ends when a disconnect signal is received, indicating that either the originating or terminating user has disconnected.

Issued: June 17, 2021 Effective: July 1, 2021

Kelly Faul, Senior Manager (T) 22000 AOL Way (T) Dulles, VA 20166 (T)

6. <u>RATES AND CHARGES (Cont'd)</u>

6.3 <u>Switched Access Services (Cont'd)</u>

6.3.2 Billing of Access Minutes

When recording originating calls over FGD with SS7 signaling, usage measurement begins with the transmission of the initial address message by the switch for direct

Trunk groups and with the receipt of an exit message by the switch for tandem Trunk groups. The measurement of originating FGD usage ends when the entry switch receives or sends a release message, whichever occurs first.

For terminating calls over FGD with SS7 signaling, the measurement of access minutes begins when the terminating recording switch receives the initial address message from the terminating End User. On directly routed Trunk groups or on tandem routed Trunk groups, the Company switch receives the initial address message and sends the indication to the Customer in the form of an answer message. The measurement of termination FGD call usage ends when the entry switch receives or sends a release message, whichever occurs first.

6.3.3 Rates and Charges

6.3.3.1 Service Implementation

Α.	Installation Charge (Per End)	
	<u>DS-1</u>	<u>DS-3</u>
	ICR	ICB

6.3.3.2 <u>Change Charges (Per Order)</u>

A. Service Date

В.	Design Changes	\$ 0.00		
C.	Expedite Charge			(T)
	On-Net Services – 100 Mbps and On-Net Services – above 100 Mb Off-Net Services*	ow	\$ 675.00 \$1,500.00 \$ 500.00	(N)(I) (N)(I)

Per Occurrence

\$ 0.00

6.3.3.3 Cancellation Charges (Per Order)

\$200.00

* Associated expedite charges imposed by the third-party provider would be applied in addition to this charge.

(N)

Issued: August 7, 2013

Kelly Faul – Regulatory Affairs Director 13865 Sunrise Valley Dr. Herndon, VA 20171 Effective: August 30, 2013

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

6. <u>RATES AND CHARGES (Cont'd)</u>

6.3 <u>Switched Access Services (Cont'd)</u>

6.3.3 Rates and Charges (Cont'd)

6.3.3.4 Switched Access

6.3.3.4.1 <u>Entrance Facility Charge</u>:

<u>Per DS1 - The rates and charges for DS1 Entrance Facilities are the charges set forth for the Company's DS1 Special Access Service as specified in Section 6.2.4.2 of this Price List.</u>

<u>Per DS3 - The rates and charges for DS3 Entrance Facilities are the charges set forth for the Company's DS3 Special Access Service as specified in Section 6.2.5.2 of this Price List.</u>

6.3.3.4.2 <u>Direct Connect Charges</u>:

	Per Access Minute of Non-8YY	Per Access Minute of 8YY	Per Access Minute
	Originating Use	Originating Use	Terminating Use
Zone 1:*	\$0.046458	Note 1*	Note 1*
Zone 2:*	\$0.046458	Note 1*	Note 1*

6.3.3.4.3 Tandem Connect Charges:

A. <u>Tandem Connect Without Tandem Switching</u>

Note 1*

	Per Access Minute of Non-8YY	Per Access Minute of 8YY	Per Access Minute
	Originating Use	Originating Use	Terminating Use
Zone 1:*	\$0.062458	Note 1*	Note 1*
Zone 2:*	\$0.062458	Note 1*	Note 1*
	В.	Tandem Connect W	ith Tandem Switching
	Per Access Minute	Per Access Minute	Per Access Minute
	of Non-8YY	of 8YY	
	of Non-8YY Originating Use	of 8YY <u>Originating Use</u>	Terminating Use

Note 1*

Zone 2:*

\$0.064927

Issued: June 15, 2022 Effective: July 1, 2022

Karl Tucker - Director (T) 22001 Loudoun County Pkwy (T) Ashburn, VA 20147 (T) **(T)**

^{*}Note 1 – 8YY Originating Usage and Terminating Usage switched access rates are billed at Company's tariffed interstate rates found in XO Communications Services, LLC FCC Tariff No. 1 Section 6.3.3 which may be viewed at www.verizon.com/tariffs/.

^{*} Rate Zones are defined in Section 1.

(R)

(R)

INTRASTATE ACCESS SERVICES PRICE LIST

6. RATES AND CHARGES (Cont'd)

6.3 <u>Switched Access Services (Cont'd)</u>

6.3.3 Rates and Charges (Cont'd)

6.3.3.5 Chargeable Optional Features

6.3.3.5.1 800 Data Base Access Service Basic Query

Zone 1**: \$0.001850 Zone 2**: \$0.002224

6.3.3.5.2 Signaling Transfer Point Access

MonthlyNon-RecurringVia ThirdPer MilePer PortPartyICBICBICB

6.3.3.6 Nonchargeable Optional Features

Supervisory Signaling \$0.00

6.3.3.7 <u>Feature Group D Optional Features</u>

6.3.3.7.1 Common Switching Optional Features

Alternate Traffic Routing	\$0.00
Automatic Number Identification	\$0.00
Cut-Through	\$0.00
Service Class Routing	\$0.00
Feature Group D with 950 Access	\$0.00
Signaling System Seven (SS7)	\$0.00
Basic Initial Address Message Delivery \$0.00	
Called Directory Number Delivery	\$0.00
Flexible Automatic Number Identification Delivery	\$0.00

6.3.4 <u>Special Construction</u>

6.3.4.1 Basis for Rates and Charges

Rates and charges for Switched Access Special Construction are the same a rates and charges for Special Access Service* and are specified in Section 6.1.1 and 6.1.2 preceding.

Issued: June 15, 2022 Effective: July 1, 2022

Karl Tucker – Director (T) 22001 Loudoun County Pkwy (T) Ashburn, VA 20147 (T)

^{*}Effective May 8, 2020, Special Access Services are no longer available to new customers. Moves, additions, or changes will no longer be permitted for existing customers.

^{**} Rate Zones are defined in Section 1.

6. RATES AND CHARGES (Cont'd)

- 6.3 <u>Switched Access Services (Cont'd)</u>
 - 6.3.4 Billing Name and Address (BNA) Service
 - 6.3.4.1 Company will upon request, provide Billing Name and Address (BNA) Service to Telecommunications Service Providers (Customer), or its authorized billing and collection agent. Telecommunications Service Providers include interexchange carriers, operator service providers, enhanced service providers and any other provider of intrastate/interstate telecommunications services. BNA provides the billing name and address when calling party dials 101XXXX or authorizes collect and/or third party calls to pay for telecommunication services. BNA Service enable telecommunications service providers and authorized billing and collection agents to seek payment of their service directly from the calling party.

In no case shall Customer or authorized billing and collection agent of a customer disclose the billing name and address information of any subscriber to any third party, except that a customer may disclose BNA. This service is offered to Customer or its authorized billing and collection agent for the following limited purposes and may not be resold or used for any other purpose:

-Billing customers for using telecommunications service of that service provider and collecting amount due.

-Verification of service orders of new customers, identification of customers who have moved to a new address, fraud prevention, and similar non-marketing purposes.

(D)

Issued: November 3, 2011

Effective: November 13, 2011
Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

NOV 1 3 2011

6. RATES AND CHARGES (Cont'd)

- 6.3 <u>Switched Access Services</u> (Cont'd)
 - 6.3.4 <u>Billing Name and Address (BNA) Service</u> (Cont'd)
 - 6.3.4.2 Obligations of the Customer
 - (A) XO will only accept BNA Service orders via email or fax. With each order for BNA Service, the customer must provide the name of the authorized individual(s) as well as the email address or fax number to which the BNA information should be sent.
 - 6.3.4.3 Rates and Charges

Billing Name and Address (Per Batch Request)

Recurring Charge

\$30.00

Billing Name and Address Found/Each \$0.30

Billing Name and Address Not Found/Each

\$0.30

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

MAR 1 3 2005

Boise, Idaho

Issued: March 3, 2005

Effective: March 13, 2005

6. RATES AND CHARGES (Cont'd)

6.4 Access Service Order Charge

Company will assess an Access Service Order Charge of \$100.00 each time a Customer requests a Carrier Identification Code to be added to a Company switch. The Customer would initiate this request with a submission of an Access Service Order to the Company.

6.5 <u>Historic Invoices</u>

The Company will furnish, upon the Customer's request, copies of invoices which were originally issued to the Customer more than thirteen months prior to the request or copies of invoices that are available on-line, but that customer does not wish to retrieve from the available on-line system. Customers can request the invoice in one of two formats: electronic copy (portable data file/.pdf) or CSV/CD of Call Detail Record. Customer will be charged based on the format requested and on a per invoice basis.

<u>Format</u>	Rate Per Invoice
Electronic	\$10.00
CSV/CD of CDR	\$25.00

6.6 Service Order Changes

6.6.1 Service Order Change Charge

Customer will be assessed a non-recurring charge when it initiates a modification of a Service Order after its acceptance by XO. If a Service Order is modified after its acceptance by XO, per-Circuit change charges will apply as follows: \$50 for any change requested by Customer within five (5) days of Service Order acceptance; \$200 for any change requested by Customer between five (5) days of Service Order acceptance and five (5) days of the projected Start of Service Date; and \$400 for any change requested by Customer within five (5) days of the projected Start of Service Date. In addition, Customer must pay all third party charges imposed as a result of the change.

6.6.2 Service Order Cancellations

Customer will be assessed a non-recurring charge for each Circuit cancelled between the date of its acceptance by XO and the projected Start of Service Date. The cancellation charges will be as follows:

Circuit Type	Cancellation Charge – Per Circuit
DS1	\$750.00, plus one month's MRC
DS3	\$1,500.00, plus one month's MRC
All other Circuit types	One month's MRC and applicable installation charges

In addition to these charges, Customer must pay all third party charges imposed as a result of the cancellation.

Issued: December 13, 2007

Kelly Faul Regulatory Affairs Director 13865 Sunrise Valley Dr. Herndon, VA 20171 Effective: January 1, 2008

(N)

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

JAN 1 - 2008

6. RATES AND CHARGES (Cont'd)

6.7 Intra-Building Moves

Customers will be assessed a non-recurring charge for requesting an existing circuit to be moved to a new end point termination within the same building as the current termination. Moves of three or fewer floors will be charges as shown below. The customer will retain all services, terms, and pricing for the moved circuit once the move has been completed. The non-recurring charges below apply to DS1 and DS3 services that do not include additional infrastructure, cabling, electronics, or other materials.

Requests for moves of more than three floors will be charged an ICB rate.

	Non-Recurring Charges		
Circuit	Initial Circuit	Each Additional Circuit	
DS1	\$550.00	\$275.00	
DS3	\$550.00	\$275.00	

6.8 Switched Access IP Direct Connect

6.8.1 Direct Connect Charges:

	Per Access Minute	Per Access Minute	Per Access Minute
	of Non-8YY	of 8YY	
	Originating Use	Originating Use	Terminating Use
Zone 1:*	\$0.046458	Note 1*	Note 1*
Zone 2:*	\$0.046458	Note 1*	Note 1*

6.8.2 Tandem Connect Charges:

A. Tandem Connect Without Tandem Switching

	Per Access Minute	Per Access Minute	Per Access Minute
	of Non-8YY	of 8YY	
	Originating Use	Originating Use	Terminating Use
Zone 1:*	\$0.062458	Note 1*	Note 1*
Zone 2:*	\$0.062458	Note 1*	Note 1*

B. Tandem Connect With Tandem Switching

	Per Access Minute	Per Access Minute	Per Access Minute
	of Non-8YY	of 8YY	
	Originating Use	Originating Use	Terminating Use
Zone 1:*	\$0.064927	Note 1*	Note 1*
Zone 2:*	\$0.064927	Note 1*	Note 1*

^{*}Note 1 – 8YY Originating Usage and Terminating Usage switched access rates are billed at Company's tariffed interstate rates found in XO Communications Services, LLC FCC Tariff No. 1 Section 6.3.3 which may be viewed at www.verizon.com/tariffs/.

Issued: June 15, 2022 Effective: July 1, 2022

Karl Tucker - Director (T) 22001 Loudoun County Pkwy (T) Ashburn, VA 20147 (T) (T)

^{*} Rate Zones are defined in Section 1.

6. RATES AND CHARGES (Cont'd)

6.9 Service Extension Charge

A Service Extension Charge will be assessed when a customer requests and XO agrees to provide services at a location beyond XO's demarcation point. This non-recurring charge will be applied to the first and each additional circuit as outlined below:

	NON-RECURRING CHARGES	
Type of Circuit	First Circuit	Each Additional Circuit
Less Than or Equal to a DS1 (1.544 Mbps)	\$ 325.00	\$ 80.00
More Than a DS1 (1.544 Mbps) and Less Than or		
Equal to a DS3 (44.736 Mbps)	\$ 550.00	\$ 550.00

Issued: October 10, 2013

Kelly Faul Regulatory Affairs Director 13865 Sunrise Valley Dr. Herndon, VA 20171 Effective: October 30, 2013

Idaho Public Utilities Commission

Office of the Secretary

ACCEPTED FOR FILING

OCT 3 0 2013

7. INTERIM INTERCONNECTION LOCAL CALL COMPLETION SERVICE (IILCC)

Interim Interconnection Local Call Completion Service will be provided to other certified LEC's for calls originating by a certified LEC within the Company's local calling area as set forth in the Company's Local Exchange Services Price List and terminating, within the same local calling area to an end user served on the Company's network. IILCC Service will be provided as a Feature Group D (FG D) trunk as described in Section 4.0 above. Rates for IILCC Service will be pursuant to individual negotiated arrangements which shall be filed with the Commission.

8. (Reserved For Future Use)

(b) (D)

Idaho Public Utilities Commission
Office of the Secretary
ACCEPTED FOR FILING

AUG 1 7 2006

Beise, Idaho

Issued: July 18, 2006

Effective: August 17, 2006